



Article

The Relationship between Corporate Social Responsibility, Global Investment, and Equity Incentives

Tong Sheng ¹, Bingquan Fang ², Xiaoqian Lu ¹, Xingheng Shi ¹, Chaohai Shen ^{1,*} and Xiaolan Zhou ^{1,*}

- ¹ Faculty of Economics and Management, East China Normal University, Shanghai 200062, China
- KoGuan School of Law, Shanghai Jiao Tong University, Shanghai 200030, China
- * Correspondence: chshen@dbm.ecnu.edu.cn (C.S.); xlzhou@jjx.ecnu.edu.cn (X.Z.)

Abstract: Listed companies have long faced difficulties in both their global investment strategies and corporate governance improvement, while they are supposed to pay more attention to their sustainable development performance. The complex linkages between these three make the choice of corporate strategy a challenge for public companies. Given the economic downturn in the postpandemic era, the challenges for listed companies are likely to be even more acute. How companies weigh the relationships between these three and how to ensure the implementation of a global investment strategy that effectively meets sustainable development are pressing challenges. Using a sample of Chinese listed companies during 2010-2018, this paper empirically examines the relationship between corporate sustainable development performance, global investment reflected by outward foreign direct investment (OFDI), and corporate governance reflected by equity incentives with econometric tools. We show the positive effects of OFDI on corporate sustainable development performance and discover the crowding-out effect of equity incentives, which challenges the view of equity motivation. These findings are robust. We further explore the heterogeneities in terms of industries and regions. We finally provide some useful implications on how to coordinate the global investment and internal equity incentives to improve corporate sustainable development performance.

Keywords: corporate sustainable development; corporate social responsibility; outward foreign direct investment; equity incentives



Citation: Sheng, T.; Fang, B.; Lu, X.; Shi, X.; Shen, C.; Zhou, X. The Relationship between Corporate Social Responsibility, Global Investment, and Equity Incentives. *Sustainability* 2022, 14, 16208. https://doi.org/10.3390/su142316208

Academic Editors: Tomasz Kijek, Aleksandra Kowalska, Arkadiusz Kijek and Anna Matras-Bolibok

Received: 25 October 2022 Accepted: 2 December 2022 Published: 5 December 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https://creativecommons.org/licenses/by/4.0/).

1. Introduction

Corporate sustainable development and corporate social responsibility (CSR) have been considered the key strategic initiative for sustainable development in the long term [1]. As a new management paradigm, corporate sustainable development requires companies to safeguard environmental, social and economic stakeholders while achieving profitable growth [2]. Great corporate sustainable development performance can not only help create value for companies [3] but also vastly enhance corporate image. Currently, many scholars focus on studying the link between corporate sustainability development and corporate social responsibility (CSR). Generally speaking, CSR is an important component of corporate sustainable development. The definition and scope of the two have a strong consistency at the corporate level.

According to the latest global CSR report published by KPMG, there has been a rapid growth in the number of companies incorporating CSR information into financial reports in the past years, and around three quarters of the companies publish social responsibility reports [4]. Compared with developed countries, less developed countries have significantly less experience in internationalization, and their CSR requirements are comparatively lax [5]. However, the Chinese government announced a new "going-out" strategy in the late 1990s. It has been noted that political transitions in government at the national level can lead to changes in economic policy. Fortunately, China's political

Sustainability **2022**, 14, 16208 2 of 27

environment is relatively stable, and the policies implemented have good continuity [6]. Therefore, the role of Chinese enterprises in the world has been attracting attention for three decades, which also requires them to take on more corporate social responsibility. Chinese companies have had to make appropriate adjustments to their original organizations in order to adapt to the changing environment under pressure from citizens, society, and NGOs in different countries. Yin and Jamali [7] note that multinational corporations (MNCs) have to meet the legitimate demands of multiple stakeholders in the countries where they operate while creating benefits for shareholders, because they may face uncertainties and costs caused by the external environment. For instance, they can leverage their expertise and professional resources to conduct public product R&D abroad [8]. MNCs can create a performance for themselves while complying with the requirements of government regulations, social norms, and culture [9].

As an important channel for the global diffusion of technology, knowledge [10], and efficiency [11], outward foreign direct investment (OFDI) can accelerate the sustainable development of enterprises in less-developed countries by transforming the industrial model to low energy consumption and low pollution [12,13]. Thanks to the reverse spillover effect, companies that make outward foreign direct investments can draw on advanced technology and management knowledge from abroad to improve their competitive advantage in their home countries and internationally [14,15]. Previous studies have established that OFDI has positive effects on multinational enterprises (MNEs) [16,17]. For example, multinational firms can acquire strategic assets through OFDI that are not available domestically, such as advanced technologies, expertise, and management upgrades. Then they can transfer these strategic assets to their home countries [18]. OFDI is a major driver for firms in emerging markets to improve their innovation capabilities and achieve sustainable development, especially for knowledge-seeking firms in developing countries [19]. After the initiation of OFDI, there is a significant increase in firms' green patents in the country, which promotes environmental performance [20]. Firms implementing OFDI can create higher-quality products [21]. However, they mainly focused on increasing productivity [16,22,23], and other aspects concerning sustainable development are rarely discussed. It has been found that OFDI can promote productivity growth of the parent company, especially the OFDI strategy of acquiring advanced technology and investment in developed countries [24]. OFDI contributes to multinational firm productivity growth by acting directly on the level of technology [25]. In the Chinese market, the implementation of OFDI by multinational firms can have a significant increase in green total factor productivity [23]. OFDI has a more significant effect on sustainable productivity for firms with developed countries as their investment destination, for firms with single branches, and for firms oriented towards overseas trade sales [9]. Technology-seeking OFDI particularly increases the productivity of multinational firms in R&D and processing [15].

In addition to focusing on the productivity dimension, some literature on measures of overall corporate sustainable development performance covers limited dimensional information. In the contribution dimension, some scholars have studied the involvement of multinational corporations with sustainable development performance, using charitable contributions and community contributions as a measure of CSR [26,27]. In the stock return dimension, some scholars have studied the relationship between the financial and social responsibility of small and medium enterprises in Vietnam using the stock performance of firms as a measure of sustainability [3]. Moreover, some scholars have used the value of the stock market as a sustainable performance of listed companies [28]. The literature mentioned does not provide a comprehensive picture of corporate sustainability. As we have stated above, corporate sustainability requires that companies achieve both non-profit and profitability targets. A single indicator may be one-sided in its representation of corporate sustainability, while a comprehensive indicator that integrates environmental, economic, social, and other aspects should be more scientific. Since there is no perfect indicator for measuring corporate sustainable development, we believe that CSR, which is similar to corporate sustainable development, is an appropriate measure. More importantly, Sustainability **2022**, 14, 16208 3 of 27

past literature on the relationship between OFDI and sustainable development has mainly focused on the national and regional levels. For example, some scholars have found direct and indirect effects of OFDI-induced reverse knowledge spillover on national sustainable development [29]. Growing OFDI can effectively reduce carbon emissions in multiple countries and provide regional environmental sustainability [30]. OFDI can improve ecological quality and promote regional economic transformation and sustainable development by improving technological innovation, upgrading industrial structure, and mitigating capital misallocation [31,32]. To our knowledge, there is little literature on the impact of OFDI on sustainable development at the firm level, and there is also a lack of analysis of the heterogeneity of OFDI's impact on corporate sustainable development across industries and regions. Therefore, we examined the relationship between corporate sustainable development and OFDI, which can effectively fill the gap in existing studies. At the same time, existing research has not found whether equity incentives have a facilitative or moderating effect on the relationship between OFDI and corporate sustainable development. Equity incentives are a key part of the growth strategy of enterprises. We believe it is essential to explore the impact of equity incentives. In this paper, our innovative choice of equity incentives as a moderating variable on OFDI and corporate sustainable development provides insights into firms' strategic choices and fills the gap in existing research.

When referring to corporate globalization strategies and corporate sustainable development performance, the impact of the board of directors and the top management team (TMT) has received particular attention [33,34]. Especially, the internal administration of MNEs is of great significance because it determines the effectiveness of the OFDI. Accordingly, as major administrators of a company, the characteristics of directors and TMT including gender [35], education background [36], and others have been widely studied [37,38]. However, to the best of our knowledge, rarely have previous studies investigated the impact of managers holding equity in the company in the circumstance of performing corporate sustainable development practices.

Agency theory suggests that the separation of ownership and management leads to agency costs. Agents may act in their personal interest to the detriment of the company's long-term interests [39]. When corporate sustainability is often treated as a long-term goal [40], agents may see it as a financial loss [41] for the sake of short-term profits under certain incentive conditions, which is not conducive to corporate sustainable development. Equity incentives are common means of mitigating agency problems. Eisenhardt points out that higher incentives align agents' efforts with the principal's desires [42]. Although it has also been noted in the literature that board shareholdings can increase the supervision of executive directors and thus avoid deviations from the interests of society [43], there is also literature that suggests a different outcome. Some researchers point out that shareholding structure and board composition affect disclosure and that an increase in outside directors reduces corporate disclosure [44,45]. Moreover, some researchers have pointed out that when the shareholding of managers is too high, it can lead to an adverse effect on the value of companies [46]. The entrenchment hypothesis suggests that excessive shareholding gives managers greater voice and management power [47]. This leads them to indulge in non-benefit-maximizing objectives [48]. Excessive power may allow managers to take the stronger side in resisting business decisions such as OFDI that are for the long-term good. Therefore, it is also essential to answer the following questions: Does the board and TMT shareholding alleviate the agency problem of corporate governance? Does higher shareholding enhance the effect of OFDI on corporate sustainable development performance?

To bridge the gaps mentioned above, this paper provides some new empirical evidence through a large number of panel data of Chinese listed companies. We find the positive impact of OFDI on comprehensive corporate sustainable development performance, with the moderating effect of internal corporate governance from the aspect of equity incentive. We contribute to the literature from the following aspects. First, we provide new insights into the spillovers of corporate globalization strategies from the perspective of

Sustainability **2022**, 14, 16208 4 of 27

corporate ecological management. We employ an accurate and comprehensive indicator for corporate sustainable development performance that takes companies that do not disclose CSR reports into consideration. To the best of our knowledge, our study is the first to adopt an accurate CSR index when examining OFDI reverse spillovers on corporate sustainable development performance. Second, we combine the internal and external corporate management approach by analyzing the relationship between the director and TMT shareholding and CSR practices. Hence, we further contribute to the literature by confirming the limitation of equity incentives, which is contrary to previous studies and general recognition. Third, we find that the multidimensional characteristics of the CEO play an important role in corporate sustainable development, thus providing some inspiration for companies' human resource management practices. Fourth, we focus on the relationship between OFDI and corporate sustainable development, which fills the gap in the existing literature. What is equally significant is that the current literature lacks an exploration of the channels between the OFDI and corporate sustainable development, and we introduce equity incentives as a moderator into the empirical study.

The rest of this paper is organized as follows. Section 2 reviews related literature and develops our hypothesis. Section 3 presents the data and methodology used in our study. Section 4 presents the results. Section 5 provides discussions. Section 6 concludes.

2. Literature Review and Hypothesis Development

2.1. Corporate Sustainable Development and CSR

According to the United Nations, sustainable development is "development which meets the needs of the present without compromising the ability of future generations to meet their own needs". It includes 17 sustainable development goals. Among them, the goal of sustainable development is the same as the general purpose of CSR at the corporate level. In addition to addressing current and future world issues, corporate sustainable development is in line with the ideal framework of CSR programs [49]. For enterprises, the role of corporate sustainability development is to create value for business and society [50] and is often discussed as a "responsibility" to society [51]. It is worth noting that CSR can play an integral role in achieving the SDGs in developing economies. Funds from the implementation of CSR can often be used to achieve corporate sustainable development goals. Simply put, at the corporate level, there is consistency between the CSR they undertake and their sustainable development goals [52]. Thus, although there are differences in the scope of CSR and corporate sustainable development, many scholars have chosen to use CSR as an aid in measuring corporate sustainability in recent years due to the limited measurement and the consistent definition of the two at the firm level [3].

Currently, there are multiple definitions of CSR. For instance, the World Business Council for Sustainable Development (1998) defines CSR as "a constant commitment of enterprises to business ethics and to contribute to economic growth, while improving the living quality of the workers and their families, as well as the local community and society." In 2001, the EU defined CSR in a Green Paper as follows: "the voluntary integration of social and environmental issues in the day-to-day business operations of an enterprise and in its interaction with its stakeholders." Although there are nuances in the definitions of CSR by different researchers, they still ultimately boil down to a few aspects: environment, shareholder rights, government, and employee rights.

Scholars have conducted research on these different aspects. First and foremost, multinational enterprises have been proven to stimulate awareness of environmental protection [53]. However, some studies have shown that while CSR can increase social welfare, it is not always good for the environment [54]. Secondly, in terms of shareholder equity, there is a positive relationship between CSR and corporate financial performance [3,55]. There is also literature that examines the combined impact of CSR on corporate performance by looking at the share price performance of companies [56]. Thirdly, government has long been widely considered an important factor in corporate behavior [57]. Gond et al. highlight the various roles that governments can play in promoting CSR in the context of

Sustainability **2022**, 14, 16208 5 of 27

wider national governance systems [58]. Finally, from the perspective of employee benefits, Singhapakdi et al. find that incongruence between employee's and firm's CSR orientation is negatively associated with both lower- and higher-order need satisfaction [59]. Golob and Podnar find that perceived internal CSR has an indirect effect on life satisfaction through its effects on job satisfaction [60].

In essence, CSR is an important part of the sustainable development issue. In the context of globalization, relevant regulations in developing countries have been changing, spreading from financial models focusing on profit only to societal models of CSR focusing on people, environment, and profit [61]. Thus, the sustainable development of companies has become a significant topic in the field of business management.

2.2. OFDI and Corporate Sustainable Development Performance

As an important carrier of foreign investment, OFDI determines the long-term development of multinational enterprises (MNEs). Various studies have assessed the positive effects of OFDI on domestic companies on multiple dimensions, including improving technological and management capacity, which in turn contributes to corporate sustainable development.

Based on international technology transfer, the increase in the total-factor productivity of corporations in the course of OFDI has been demonstrated [13,23]. Due to the existence of reverse technology spillover [14,18], domestic companies can increase their technological innovation and productivity through investment activities abroad [62]. Chen et al. [63] use R&D spending as a proxy for MNEs' technological capability and find that MNEs in emerging markets that have subsidiaries in host developed markets exhibit stronger technological capabilities in the home country. In particular, in the process of OFDI, multinational companies can make up for the original technical disadvantages through M&A to improve their technology competitiveness [64]. Additionally, cooperation with local enterprises has expanded access to R&D resources and knowledge [65]. Domestic innovation performance is also positively related to OFDI in developed countries [66]. However, the impact of OFDI on green technology and production has not reached a consensus. Zhou et al. find that the impact of OFDI on the green economy of Chinese provinces is heavily dependent on provincial heterogeneity, with most provinces struggling to benefit from OFDI [67]. In addition, the impact of OFDI on domestic R&D expenditure may be uncertain because of complementary and substitution effects [68].

On the other hand, the performance of enterprises not only depends on the technology but also on management ability. Managers learn through observing, incorporating, and sharing others' experiences [69]. As latecomers to international markets, companies from emerging markets often use internationalization as "a springboard to acquire strategic resources" [70]. Learning about management and marketing is one of their motivations for undertaking OFDI [71]. Many Chinese companies engage in exploratory OFDI to seek to develop new management knowledge [72]. Through outward investment, managers are able to learn from advanced international governance experience, thereby improving corporate governance. In the OFDI process, managers of MNEs in emerging economies have been exposed to new forms of organizational know-how about governing firm activities [73]. Additionally, companies that expand abroad are often perceived as outsiders by their host countries [74]. Due to isomorphic pressure, they tend to learn managerial capabilities from local companies [75], thereby improving corporate governance.

At the same time, technology and management capabilities have proven to be two important channels through which OFDI can contribute to CSR. First, the revolution in technology is essential for multiple dimensions of corporate sustainability. From the perspective of environmental value, technological innovation has become fundamental to environmental protection and sustainable economic development [76]. It is a new means for the clean and optimal use of scarce resources, thus promoting environmental and ecological sustainability [77]. China's OFDI increases domestic environmental pollution by expanding economic scale, while the reverse technology spillover effect from OFDI

Sustainability **2022**, 14, 16208 6 of 27

raises domestic technology levels and improves industrial structure, thereby reducing domestic environmental pollution [78]. From the perspective of social value, the application of advanced technology, especially industrial robotics, ensures employee safety, thus improving employee welfare [79]. From the perspective of economic value, perceived technological attributes influence consumer brand engagement, leading to corporates' utilitarian benefits from brand-related interactions [80]. Moreover, Shahzad et al. state that green innovative technology affects all dimensions of a firm's sustainable performance [81]. The use of technology also has a positive effect on CSR disclosure [82].

Second, in terms of managerial capability, firms can enhance their management capabilities through OFDI and thus improve their investment efficiency [83], reduce production and information costs [84], and improve their profit margins [85]. Wang and Hu find that the more experience a company has in OFDI and more exposed it is to the international environment, the more likely it is to adopt the prevailing international management norm of CSR [86]. Bhaumik state that foreign investments by emerging economy firms lead to the upgrade of their management capabilities [73]. Strong corporate governance has in turn encouraged more professional CSR activities [87] or increased CSR disclosures [88]. Therefore, companies can overcome their internal management weakness through OFDI [89] in order to better align themselves with global policies including CSR standards that promote trade and investment liberalization.

Based on the above discussion, we hypothesize that:

Hypothesis 1. OFDI can promote corporate sustainable development performance.

2.3. Equity-Based Incentives of Corporate Internal Governance

Because the board of directors and TMT (top management team) have a strong lead in the company's strategic decisions, their impact has been widely discussed when referring to CSR activities. The majority of prior studies have found that OFDI has a positive effect on sustainable development in homeland [90–92]. However, few studies have focused on the moderating effect of equity incentives in the process.

Research shows that the board of directors' and TMT's personal characteristics affect CSR performance. For instance, a higher percentage of women on the board of directors helps to improve the quality of CSR disclosure [93]. Especially, Saeed et al. prove that female executives' values can translate into environmental strategy, which will promote corporate sustainable development performance [94]. Additionally, narcissism as a personality trait of the CEO is positively related to aggregate CSR [95]. Overall, the role of the board of directors and TMT in CSR performance is well documented.

In addition to the personal characteristics of executives and directors that affect CSR performance, the way they are motivated at a particular level also affects the way they behave in corporate governance [96]. For example, the CSR of SOEs was significantly lower than that of non-SOEs due to the lack of alignment between the interests of controlling shareholders and their companies. Equity incentive plans in SOEs are able to better facilitate CSR performance after SOE shareholders are allowed to exchange their stock holdings [97]. The compensation and tenure offered by firms to their TMTs are often linked to the achievement of corporate social performance [98]. However, there is still relatively little discussion about the board and TMT incentive approach in the context of CSR.

As an important part of the enterprise incentive mechanism, there has been a lot of interest in equity incentives for directors and TMT. It is a much-debated question whether equity incentive has positive effects on corporate management, but we have to focus on this issue because management capacity is important in the mechanism by which OFDI affects CSR, as already discussed above. Corporate governance is originated from agency theory [99]. Separating ownership and the right of management causes differing interests between shareholders and managers. Incentives serve to align agents' efforts with the principal's desires [42]. For example, in order to reduce adverse selection and moral hazard caused by agency problems, appropriate equity incentives can ensure the maximization of

Sustainability **2022**, 14, 16208 7 of 27

shareholders' interests under certain conditions [99]. Equity incentives are long-term in nature. Meanwhile, agency problems tend to focus on short-term interests. A number of scholars have found that equity incentives can converge short-term interests to long-term interests, thus reducing adverse selection and moral hazard [100–102]. Equity incentives also contribute to earnings management to increase the value of the shares [103]. A part of the research shows that equity-based incentives help to improve companies' financial performance [104].

While the role of equity incentives in the classic literature is more about aligning the interests of managers with those of shareholders, aligning equity incentives with a company's corporate responsibility policy is a greater challenge. Although a large body of past literature illustrates the beneficial effects of equity incentives on corporations, most studies have not focused on specific corporate goals, such as CSR. This is problematic because equity incentives do not guarantee that the interests of agents and principals will be aligned under any corporate goals. Meanwhile, the agent in the enterprise is not a single. Dual-agency theory proposes a trilateral relationship between the principals, the directors, and the TMT on the basis of original. Therefore, it is necessary to discuss the two different sets of agents, the board of directors and the TMT, separately.

(1) The shareholding of directors and CSR

Generally, the board of directors is the executive body of the shareholders' meeting with a legal position to effectively regulate the executive team [105]. Their main role in corporate governance is to set the basic values and standards for the company, to perform the duties of resource provision and supervision [106], and to ensure the smooth implementation of all macro operations. It has been argued that the board's role is not only to set macro business goals but also to achieve CSR in order to achieve good corporate governance and reduce conflicts of interest between managers and non-investment stakeholders [107]. Mackenzie pointed out that internal organizational incentives have an important role in the fulfillment of CSR, especially the development of board goals and incentives in achieving CSR [108]. Giving appropriate equity incentives to the board of directors who monitor the CEO is related to the alignment of the goals of ownership and operation, facilitating the achievement of long-term corporate goals such as CSR [99].

By reviewing the literature, we find that past studies have generally concluded that OFDI has a catalytic effect on the sustainable development of firms, but have not considered the role played by the equity share of directors. We argue that board equity share has the potential to moderate the relationship between OFDI and corporate sustainability for the following reasons.

First, an increase in board shareholding can have an impact on the effectiveness of OFDI by influencing disclosure, which in this case itself includes CSR disclosure. It has been shown that paying directors higher compensation (or additional compensation) is highly correlated with higher information asymmetry and stronger structural management power is detrimental to the information environment of the firm [109]. Mahoney and Roberts explain the relationship between CSR disclosure and director ownership. They find that when boards of directors increase their control over the firm through, for example, equity incentives, they tend to reduce CSR disclosure to attract investment [110]. Akhtaruddin and Haron also find a weak negative relationship between the board's equity holdings and voluntary disclosure [111]. In fact, lower-quality CSR disclosure implies lower actual CSR performance [112,113]. Furthermore, it is documented that the lack of information disclosure due to rising board equity is likely to reduce investment efficiency. Higher corporate disclosure can motivate managers to act in the best interest of shareholders, thus increasing the efficiency of capital investment [114,115]. Especially for OFDI, as a risky venture investment, information disclosure plays an important role in risk sharing [116]. Therefore, the degree of information disclosure by the board of directors is more important for the efficiency of OFDI. Better information disclosure helps managers to comprehensively consider the investment ability of the enterprise and improve the effectiveness of OFDI implementation, thus helping to enhance the sustainable development of the enterprise.

Sustainability **2022**, 14, 16208 8 of 27

In addition, the increase in board shareholding may also affect the relationship between OFDI and corporate sustainability by reducing regulatory efforts. The board of directors is not only the head of executives, but also the agent of major shareholders. By implementing equity incentives for directors, there is a risk of creating secondary agency problems between directors and shareholders [106]. When the interests of directors and shareholders are inconsistent in the implementation of OFDI in enterprises, some directors may relax effective supervision out of their own interests [106], which brings negative impact on OFDI and thus affects the fulfillment of CSR.

In conclusion, boards of directors lead the development of corporate strategies including CSR strategies, and their incentive structures influence the deployment and implementation of the strategies [117]. The behavior of the board of directors in corporate macro governance can have an impact on the CSR objectives of the firm. Moreover, board shareholding reduces the quality of information disclosure, which is associated with investment efficiency. Increasing in board shareholding may bring about secondary agency problems, which may weaken the board's ability to regulate effectively. The board's shareholding has the potential to have a moderating effect between OFDI and CSR through the channels described above. On this basis, we propose the following hypothesis.

Hypothesis 2. The shareholding of directors moderates the relationship between OFDI and corporate sustainable development performance.

(2) The shareholding of top management team and CSR

The TMT and the board of directors are two key teams in improving corporate performance. The difference is that TMT is responsible for taking strategic action, while board members fulfil their resource provision and oversight roles [118]. TMT's primary fiduciary responsibility is to manage the company in a manner that promotes the interests of shareholders [119]. Therefore, equity incentives to align the interests of TMT members with those of shareholders are a common means of solving agency problems [120].

We previously discussed the impact of OFDI on CSR performance and also summarized the initial application of equity incentives in corporate governance. It is worth noting that these factors are not orthogonal to each other, but interact with each other. In other words, the effectiveness of OFDI on CSR performance may depend on the structure of equity incentives within the firm for TMT, i.e., the extent to which they are constrained by the firm's stock returns [121]. CSR is a strategic means to obtain long-term profitability and help firms build good reputation [40]. However, some literature proves a negative correlation between CSR investment and financial performance [41], especially for short-term stock returns. What this means is that when the level of TMT holdings reaches a certain level, they may possibly pay less attention to corporate reputation, and regard CSR investment as a financial loss as Crisóstomo et al. have proved [41]. This also implies that TMT tends to pay more attention to the short-term financial performance of the firm in order to maximize their benefits. Benmelech et al. point that equity incentives induce managers to conceal bad news about future growth options, and this bad news hoarding by managers leads to the overvaluation of a firm's stock [122]. Meanwhile, equity incentive-based compensation structure will enable TMT to maximize their own profits when making decisions about dividends, which is not conducive to the overall development of the company [39].

Specifically in terms of a company's globalization strategy, equity incentives may cause TMT to deviate from the norm in the execution of OFDI. Institutional scholars have demonstrated that companies encounter legitimacy pressures from host countries when expanding abroad [123,124]. Furthermore, it has been well documented that CSR is an important way for companies to gain legitimacy [125,126]. The efforts of companies seeking to conform to host countries' legitimacy norms are consistent with CSR strategies to some extent (e.g., carbon trading restrictions [127] and appeals of stakeholders [128]) [74]. However, the pursuit of legitimacy often means increased business risk for companies. For example, companies that conform to legitimacy may face greater acquisition costs when acquiring across borders [129]. Managers must balance legitimacy risk with business risk

Sustainability **2022**, 14, 16208 9 of 27

when executing OFDI [130,131]. Therefore, TMT's concerns about protecting equity wealth under equity incentives can make them less likely to conform to local legality norms [132], leading them to abandon legitimacy goals, such as CSR goals.

In sum, equity incentives may further constrain the role of OFDI for CSR by influencing the operation behavior of TMT. As the executor of the business operations, TMT likewise dominates the specific strategies of OFDI [133,134]. Incentive may lead TMT to engage in OFDI with biases, such as pursuing short-term stock profits at the expense of long-term benefits, corporate reputation, or legitimacy goals, thereby mitigating the positive impact of OFDI on CSR. Therefore, equity incentives for TMT may not have positive effects on CSR performance as expected and probably even have a crowding-out effect on the positive regulation of OFDI. On this basis, we propose the following hypothesis:

Hypothesis 3. The shareholding of top management team moderates the relationship between OFDI and corporate sustainable development performance.

3. Data and Sample Overview

3.1. Sample Selection and Data Source

We use listed companies in China as research samples. CSR data used in this study come from Hexun database. Hexun scores by the CSR reports and annual financial reports issued by all the Shanghai Stock Exchange (SSE) and Shenzhen Stock Exchange (SZSE) listed companies. Based on the specialized quantitative model, Hexun measures comprehensive CSR performance of companies including both the detail of the CSR information disclosure and the degrees of taking social responsibility. Considering the availability of data, our samples are from 2010 to 2018. Data on FDI and corporate governance are from the China Stock Market and Accounting Research (CSMAR) database. Excluding observations with abnormal and missing values, we obtained 3250 listed companies, covering 31 provinces (autonomous regions and municipalities) and 79 industries in China. Table 1 presents the distribution of our sample.

Table 1. Sample distribution.

| (1) | By Province | ce | (2) By Indu | | |
|-------------------|-------------|----------------------|-----------------------------------|--------|----------------------|
| Province | Number | % of Total Sample | Industry | Number | % of Total Sample |
| Anhui | 639 | 3.3 | Hotels and restaurants | 92 | 0.5 |
| Beijing | 1669 | 8.6 | Warehousing | 58 | 0.3 |
| Fujian | 664 | 3.4 | Manufacturing | 7894 | 41.0 |
| Gansu | 162 | 0.8 | Health care | 1138 | 5.9 |
| Guangdong | 3310 | 17.1 | Mining | 575 | 3.0 |
| Guangxi | 232 | 1.2 | Real estate | 1290 | 6.7 |
| Guizhou | 123 | 0.6 | Wholesale and retail | 891 | 4.6 |
| Hainan | 162 | 0.8 | Technology | 1808 | 9.4 |
| Hebei | 347 | 1.8 | Transport | 485 | 2.5 |
| Henan | 547 | 2.8 | Finance | 390 | 2.0 |
| Heilongjiang | 172 | 0.8 | Culture and sports | 313 | 1.6 |
| Hubei | 523 | 2.7 | Agriculture, fishing, forestry | 325 | 1.7 |
| Hunan | 576 | 2.9 | Textiles and clothing | 575 | 3.0 |
| Jilin | 237 | 1.2 | Metal smelting | 597 | 3.1 |
| Jiangsu | 2088 | 10.8 | Professional technique service | 145 | 0.8 |
| Jiangxi | 237 | 1.2 | Environmental governance | 110 | 0.6 |
| Liaoning | 482 | 2.5 | Information technology | 1000 | 5.2 |
| Inner Mongolia | 136 | 0.7 | Electricity, gas and water supply | 506 | 2.6 |
| Ningxia | 89 | 0.4 | Food processing | 559 | 2.9 |
| Qinghai | 60 | 0.3 | Public administration | 514 | 2.7 |

Sustainability **2022**, 14, 16208 10 of 27

| — | 1 1 | | - | | | | |
|----------|-----|---|---|-----|------------|----|--|
| Ta | n | 0 | | - 1 | O 1 | иt | |
| | | | | | | | |

| (1 |) By Provinc | ce | (2) By Industry | | |
|-----------|--------------|---------------------------------|-----------------|----------------------|--------|
| Province | Number | iber % of Total Industry Number | | % of Total Sample | |
| Shandong | 1175 | 6.0 | Total | 19,265 | 100.00 |
| Shanxi | 191 | 0.9 | (3) By | Year | |
| Shanghai | 1228 | 6.3 | 2010 | 1601 | 8.3 |
| Shaanxi | 252 | 1.3 | 2011 | 1840 | 9.5 |
| Sichuan | 708 | 3.6 | 2012 | 1913 | 9.9 |
| Tianjin | 269 | 1.3 | 2013 | 1922 | 9.9 |
| Tibet | 80 | 0.4 | 2014 | 2001 | 10.3 |
| Xinjiang | 262 | 1.3 | 2015 | 2181 | 11.3 |
| Yunnan | 216 | 1.1 | 2016 | 2376 | 12.3 |
| Zhejiang | 2185 | 11.3 | 2017 | 2017 2698 | |
| Chongqing | 244 | 1.2 | 2018 | 2733 | 14.1 |
| Total | 19,265 | 100.00 | Total | 19,265 | 100.00 |

Note: The classification of 79 industries is based on the industry classification standards set by China Securities Regulatory Commission (CSRC) in 2012. Due to space limitations, some consolidations have been made in the table.

3.2. Econometric Model

We use unbalanced panel regression analysis to identify the OFDI reverse effects on CSR. Considering differences in policy and other factors across provinces and industries, we use an econometric model with three-way fixed effects. The econometric models are as follows:

$$CSR_{ijpt} = \alpha_0 + \alpha_1 \ OFDI_d_{ijpt-1} + \alpha_2 \ X + u_i + u_p + u_t + \varepsilon_{ijpt}$$
 (1)

$$CSR_{ijpt} = \beta_0 + \beta_1 \ OFDI_d_{ijpt-1} + \beta_2 \ OFDI_d_{ijpt-1} \times dir_share + \beta_3 X + u_j + u_p + u_t + \varepsilon_{ijpt} \quad (2)$$

$$CSR_{iivt} = \beta_0 + \beta_1 \ OFDI_d_{iivt-1} + \beta_2 \ OFDI_d_{iivt-1} \times tmt_share + \beta_3 X + u_i + u_p + u_t + \varepsilon_{iivt}$$
 (3)

where the subscript i, j, p and t denote firm, industry, province, and year, respectively. CSR refers to companies' performance of social responsibility. Two interactions (dir_share and tmt_share) are added into model (2) and (3). X is a vector of the control variables, including age, size and roe of a company in a certain year, and CEOs' gender, age, study and work experience. u_j , u_p and u_t represent industry, province and year fixed effects, respectively.

We use a 1-year lag for OFDI as the independent variable, which helps eliminate the reverse causality problem. In order to reduce omitted variable bias, we add a comprehensive set of control variables including corporate and CEO characteristics.

3.3. Measures of Variables

Corporate Sustainable Development Performance: CSR

Data on CSR in this study are from the Hexun database. CSR scores for Chinese listed companies from Hexun focus more on the CSR performance of companies than disclosure quality, which makes them more suitable for the purpose of our study. The scores are based on these companies' social responsibility reports and financial statements. The professional evaluation index system includes five dimensions, including shareholder equity, employee benefits, consumer rights, environmental protection and public benefit. In addition, it consists of 13 second-level indexes and 37 third-level indexes. The focus is on the five dimensions, which vary among different industries, and different weights are set for each dimension. Therefore, it eliminates the unscientific nature of CSR scoring to some extent. Detailed information and specific measurement indicators are provided in Table 2.

Sustainability **2022**, 14, 16208

Table 2. CSR components.

| | Level-1 Sub-Indicators | Level-2 Sub-Indicators | Level-3 Sub-Indicators |
|--------|---|------------------------------------|---|
| | | Earnings 10% | Return on equity (2%) Return in total assets (2%) Profit margin of main business (2%) Rate of return on cost (1%) Earnings per share (2%) Undistributed profit per share (1%) |
| | Shareholder (30%) | Solvency 3% | Quick ratio (0.5%) Current ratio (0.5%) Cash ratio (0.5%) Equity ratio (0.5%) Asset-liability ratio (1%) |
| | | Returns to shareholder 8% | Ratio of dividends to equity (2%) Dividend of pay-out ratio (3%) Ratio of dividends to distributable (3%) |
| | | Credit approval 5% | Number of penalties imposed by the exchange on the company and relevant responsible persons (5%) |
| | | Innovation 4% | Product development expenditure (1%) Technological innovation concept (1%) Number of technological innovation projects (2%) |
| CSR | Employee (15% in common industries, 10% in consumption industries) | Performance 5% | Per capita income employees (4%) Employee training (1%) |
| (100%) | | Safety 5% | Safety inspection (2%) Safety training (3%) |
| | | Caring for employees 5% | Employee caring consciousness (1%) List of members of caring for employees (2%) Consolation money for employees (2%) |
| | Consumer (15% in common | Product quality 7% | Quality management awareness (3%) Certificate of quality management system (4%) |
| | industries, 20% in consumption industries) | After-sales service 3% | Customer satisfaction survey (3%) |
| | | Integrity 5% | Fair competition among suppliers (3%) Anti-bribery training (2%) |
| | Environment (20% in common industries, 30% in manufacturing industries, 10% in service industries) | Environmental management 20% | Environmental protection consciousnes (2%) Environmental management system certification (3%) Investment in environmental protection (5%) Number of pollutant discharge types (5%) Number of energy-saving measure types (5%) |
| | Public (20% in common industries, 10% in manufacturing industries, 30% in service industries) | Contribution value 20% | Ratio of income tax to total profits (10% Public donation amount (10%) |

Sustainability **2022**, 14, 16208 12 of 27

As the party having management rights, the board of directors and TMT are worth paying special attention to in agency problems. According to the previous hypothesis, we use the director (*dir_share*) and TMT shareholding (*tmt_share*) as moderators. As the board of directors and TMT are important players in corporate governance, we believe that adopting the shareholding ratio of the board of directors and TMT can well reflect the equity incentive policy and structure of the enterprise, thus making the mechanism of the role between OFDI and CSR clearer.

We select control variables from two perspectives: characteristics of CEO and companies. As the formulator and implementer of corporate strategy, CEOs need to continuously make strategic design and adjustment according to the development of companies. It has been proved that the characteristics of CEOs exert important impacts on companies [93–95,135]. Hence, at the level of CEOs' characteristics, we use CEOs' gender (male), age (CEO_age), education level (edu), and the overseas employment experience (forgn_exp) as control variables. At the level of firm characteristics, we choose the size, age, and return on equity as control variables. Previous studies have proved that foreign direct investment (FDI) has spillovers on companies in host countries [136]. Therefore, we use FDI (FDI_d) as a control variable. Detailed variable definitions in our study are provided in Table 3. Table 4 provides the descriptive statistics of the variables. In addition, the small correlation coefficients of the variables indicate that there is no high correlation among variables, which avoids the problem of multicollinearity, and validates the rationality of the model. The correlation matrix is summarized in Appendix A Table A1.

Table 3. Variable definitions.

| Variable | Definition |
|-------------|--|
| CSR | A comprehensive indicator based on corporate responsibility for the shareholders, the consumers, the employees, the public and the environment. Detailed information is attached in Table 2. |
| OFDI_d | A dummy variable that equals 1 if an enterprise is engaged in outward foreign direct investment, and 0 otherwise. |
| | Moderator Variable |
| dir_share | Shares held by the board of directors. |
| tmt_share | Shares held by the top management team. |
| | Control Variable |
| FDI_d | A dummy variable that equals 1 if an enterprise gains foreign direct investment, and 0 otherwise. |
| male | A dummy variable that equals 1 if the CEO is male, and 0 otherwise. |
| CEO_age | The age of the CEO in a certain year. |
| edu | A dummy variable that equals 1 if the CEO has obtained a bachelor degree or higher, and 0 otherwise. |
| forgn_exp | A dummy variable that equals 1 if the CEO has experience working abroad, and 0 otherwise. |
| conc | Ownership concentration. Measured by share proportion of the top 3 or 5 largest shareholders. |
| size_assets | Measured by a firm's total assets for the year. |
| size_income | Measured by a firm's operating income for the year. |
| firm_age | The number of years since a firm was established in a certain year. |
| roe | Return on equity. |
| roa | Return on assets. |

Sustainability **2022**, 14, 16208 13 of 27

| Table 4. | Summary | statistics. |
|----------|---------|-------------|
|----------|---------|-------------|

| Variables | Mean | SD | Min | Median | Max | Unit |
|-------------|--------|---------|----------|--------|------------|---------|
| CSR | 25.247 | 16.411 | -18.450 | 22.370 | 87.870 | - |
| OFDI_d | 0.034 | 0.180 | 0 | 0 | 1 | - |
| dir_share | 0.150 | 0.207 | 0 | 0.007 | 0.892 | - |
| tmt_share | 0.085 | 0.154 | 0 | 0.002 | 0.843 | - |
| FDI_d | 0.177 | 0.382 | 0 | 0 | 1 | - |
| male | 0.942 | 0.234 | 0 | 1 | 1 | - |
| CEO_age | 52.499 | 7.568 | 27 | 52 | 85 | year |
| edu | 0.862 | 0.345 | 0 | 1 | 1 | - |
| forgn_exp | 0.050 | 0.218 | 0 | 0 | 1 | - |
| conc_top3 | 49.619 | 15.973 | 0.565 | 49.313 | 98.290 | % |
| conc_top5 | 54.499 | 15.885 | 0.811 | 54.904 | 98.469 | % |
| size_assets | 66.349 | 786.821 | 0.003 | 2.545 | 22,786.910 | billion |
| size_income | 10.579 | 79.749 | 0 | 1.500 | 2900 | billion |
| firm_age | 15.632 | 5.773 | 0 | 15 | 51 | year |
| roe | 0.087 | 5.580 | -176.380 | 0.070 | 713.204 | - |
| roa | 0.056 | 0.894 | -48.307 | 0.053 | 108.352 | - |

Note: To calculate the mean and SD, we use the original data here without any transformation. CSR data are from the Hexun database. Data on FDI and corporate governance are from China Stock Market and Accounting Research (CSMAR). CSMAR is a series of advanced professional financial and economic databases designed and developed to meet the needs of financial and economic analysis and research in China.

4. Results and Analysis

4.1. The Impact of OFDI on Corporate Sustainable Development Performance

Table 5 presents the regression results of OFDI spillovers on corporate sustainable development performance. As shown in Column (1), the coefficient on OFDI_d is statistically significant at the 1% level with a coefficient of 1.882, suggesting that outward foreign direct investment has significantly positive effects on corporate sustainable development performance. Thus, companies' overseas investment can be regarded as an effective channel to improve sustainability. This is probably because in the process of foreign trade, the demonstration of production skills and the transfer of tacit knowledge will lead to strengthening the awareness of environmental protection and sustainable development. Therefore, some short-sighted practices will no longer be adopted to improve companies' short-term profits. When we add interactions to the regression model in Columns (2) and (3), we find that the negative interaction effects of directors and TMT shareholding are also statistically significant, and their coefficient values are -10.784 and -14.025, respectively. Specifically, the positive impact of OFDI on corporate sustainable development is moderated by 10.784% and 14.025% when the shareholding of the board and TMT increases by 1%, respectively. Such results illustrate that equity incentives do not drive sustainable growth and that firms need to focus more on other incentives when climbing the ladder of corporate sustainable development. This finding provides new insights into corporate operations and decision making. We also use OLS with robust standard error, and the conclusions are further confirmed. It is shown in Figure 1 that while improving sustainability through OFDI, equity incentives as a traditional practice do not work well and weakens the positive effect of OFDI on CSR, thus approaches to corporate internal governance should be adjusted.

As for control variables, all of the CEO characteristics in this study have significant effects on promoting corporate sustainable development performance. First, CEOs' work experience in foreign countries (forgn_exp) will greatly help promote corporate sustainable development performance. The results reveal that they have more opportunities to understand the connotation of corporate sustainable development and pay more attention to the potential impact of CSR-related activities from the long-term standpoint. Second, CEOs who are older (CEO_age) and better educated (edu) are more receptive to and usually have more experience in solving problems encountered by companies in the process of corporate sustainable development reform. Third, male CEOs (male) perform better in taking corporate social responsibility.

Sustainability 2022, 14, 16208 14 of 27

Table 5. The impact of OFDI on corporate sustainable development performance.

| Variables | (1) | (2) | (3) | (4) | (5) | (6) |
|----------------------|----------------------|------------------------|------------------------|------------------|-----------------------|------------------------|
| | FE | FE | FE | OLS | OLS | OLS |
| OFDI_d | 1.882 *** | 3.516 *** | 2.929 *** | 1.396 * | 2.673 *** | 2.190 ** |
| | (0.657) | (0.821) | (0.746) | (0.749) | (1.023) | (0.903) |
| dir_share | (0.007) | -1.872 *** (0.714) | (0.7 ±0) | (0.747) | -5.678 *** (0.634) | (0.203) |
| tmt_share | | | -2.615 *** (0.897) | | | -6.204 *** (0.737) |
| OFDI_d ×dir_share | | -10.784 *** (3.146) | | | -8.903 *** (3.195) | |
| OFDI_d ×tmt_share | | | -14.025 *** (4.447) | | | -11.680 *** (3.333) |
| FDI_d | 1.442 *** | 1.215 *** | 1.250 *** | 5.004 *** | 4.395 *** | 4.617 *** |
| | (0.468) | (0.472) | (0.471) | (0.415) | (0.419) | (0.417) |
| male | 1.091 ** | 0.938 * | 1.066 * | 1.643 *** | 1.336 ** | 1.612 *** |
| | (0.552) | (0.553) | (0.552) | (0.553) | (0.553) | (0.551) |
| CEO_age | 0.104 *** | 0.091 *** | 0.095 *** | 0.034 * | -0.004 | 0.012 |
| | (0.019) | (0.020) | (0.019) | (0.019) | (0.020) | (0.020) |
| edu | 1.537 *** | 1.387 *** | 1.414 *** | 2.451 *** | 2.097 *** | 2.230 *** |
| | (0.394) | (0.397) | (0.396) | (0.355) | (0.356) | (0.355) |
| forgn_exp | 1.422 ** | 1.458 *** | 1.451 *** | -0.158 | 0.066 | 0.024 |
| | (0.557) | (0.557) | (0.557) | (0.578) | (0.577) | (0.576) |
| conc_top3 | 0.148 *** | 0.151 *** | 0.151 *** | 0.154 *** | 0.160 *** | 0.159 *** |
| | (0.009) | (0.009) | (0.009) | (0.010) | (0.010) | (0.010) |
| size_assets | -0.0006 ** | -0.0006 ** | -0.0006 | 0.0013 *** | 0.0013 *** | 0.0013 *** |
| | (0.0002) | (0.0002) | (0.0002) | (0.0003) | (0.0002) | (0.0003) |
| firm_age | 0.106 *** (0.027) | 0.092 *** (0.027) | 0.097 *** (0.027) | 0.015 (0.025) | -0.027 (0.025) | -0.011 (0.025) |
| roe | 0.008 | 0.008 | 0.008 | 0.013 | 0.013 | 0.013 |
| | (0.019) | (0.019) | (0.019) | (0.033) | (0.034) | (0.033) |
| Constant | 7.809 *** | 9.206 *** | 8.679 *** | 10.646 *** | 14.711 *** | 12.866 *** |
| | (1.314) | (1.382) | (1.339) | (1.290) | (1.354) | (1.314) |
| Time effect | Yes | Yes | Yes | - | - | - |
| Province effect | Yes | Yes | Yes | - | - | - |
| Industry effect | Yes | Yes | Yes | - | - | - |
| N | 13,465 | 13,465 | 13,465 | 13,465 | 13,465 | 13,465 |
| Adj. R2 | 0.195 | 0.196 | 0.196 | 0.056 | 0.062 | 0.061 |

Note: ***, ** and * indicate statistical significance at the 1%, 5% and 10% levels, respectively.

4.2. Robustness Checks

We perform the following robustness checks:

First, due to the natural differences among different types of firms, financial firms usually take social responsibility in a limited way without environmental protection by adopting new technologies. Thus, there is not much room to work with corporate sustainable development practices in the finance industry. Therefore, we exclude observations of finance companies as a robustness check. The sample size is smaller, and the spillovers of OFDI are still significant. Shareholdings of directors and TMT weaken. As shown in Table 6 from Columns (1) to (3), our hypotheses are further supported.

Sustainability **2022**, 14, 16208

(a) Interaction effects of director shareholding on CSR (b) Interaction effects of supervisor shareholding on CSR 12 12 11 10 10 Fewer shares held by the board of directors Fewer shares held by the board of directors CSR More shares held by the board of directors ---- More shares held by the board of directors 2 6 0 without OFDI with OFDI without OFDI with OFDI

Figure 1. A graphical representation of moderating effects.

Table 6. Robustness checks by eliminating samples.

| Variables | (1) | (2) | (3) | (4) | (5) | (6) |
|----------------------|-----------|------------------------|------------------------|-----------|------------------------|------------------------|
| OFDI_d | 2.028 *** | 3.692 *** | 3.107 *** | 2.028 *** | 3.692 *** | 3.107 *** |
| | (0.660) | (0.827) | (0.751) | (0.700) | (0.924) | (0.832) |
| dir_share | | -1.246 * (0.718) | | | -1.246 * (0.640) | |
| tmt_share | | | -1.904 ** (0.875) | | | -1.904 ** (0.756) |
| OFDI_d ×dir_share | | -10.800 *** (3.154) | | | -10.800 *** (3.133) | |
| OFDI_d ×tmt_share | | | -14.108 *** (4.439) | | | -14.108 *** (3.564) |
| FDI_d | 0.886 * | 0.724 | 0.742 | 0.886 * | 0.724 | 0.742 |
| | (0.481) | (0.485) | (0.483) | (0.509) | (0.514) | (0.512) |
| male | 0.900 | 0.782 | 0.884 | 0.900 * | 0.782 | 0.884 |
| | (0.557) | (0.558) | (0.557) | (0.547) | (0.548) | (0.547) |
| CEO_age | 0.087 *** | 0.078 *** | 0.081 *** | 0.087 *** | 0.078 *** | 0.081 *** |
| | (0.019) | (0.020) | (0.019) | (0.020) | (0.020) | (0.020) |
| edu | 1.439 *** | 1.333 *** | 1.347 *** | 1.439 *** | 1.333 *** | 1.347 *** |
| | (0.394) | (0.396) | (0.395) | (0.354) | (0.356) | (0.356) |
| forgn_exp | 1.378 ** | 1.403 ** | 1.397 *** | 1.378 ** | 1.403 ** | 1.397 ** |
| | (0.558) | (0.558) | (0.558) | (0.550) | (0.549) | (0.549) |
| conc_top3 | 0.141 *** | 0.143 *** | 0.143 *** | 0.141 *** | 0.143 *** | 0.143 *** |
| | (0.009) | (0.009) | (0.009) | (0.009) | (0.010) | (0.010) |
| size_assets | 0.032 *** | 0.032 *** | 0.032 *** | 0.032 *** | 0.032 *** | 0.032 *** |
| | (0.003) | (0.003) | (0.003) | (0.005) | (0.005) | (0.005) |
| firm_age | 0.123 *** | 0.112 *** | 0.116 *** | 0.123 *** | 0.112 *** | 0.116 *** |
| | (0.027) | (0.028) | (0.028) | (0.027) | (0.027) | (0.027) |
| roe | 0.008 | 0.008 | 0.008 | 0.008 | 0.008 | 0.008 |
| | (0.019) | (0.019) | (0.019) | (0.028) | (0.029) | (0.028) |
| Constant | 8.177 *** | 9.179 *** | 8.824 *** | 8.177 *** | 9.179 *** | 8.824 *** |
| | (1.323) | (1.389) | (1.346) | (1.350) | (1.379) | (1.358) |
| Time effect | Yes | Yes | Yes | Yes | Yes | Yes |

Sustainability **2022**, 14, 16208 16 of 27

Table 6. Cont.

| Variables | (1) | (2) | (3) | (4) | (5) | (6) |
|--------------------|-----------------|-----------------|-----------------|-----------------|-----------------|-----------------|
| Province effect | Yes | Yes | Yes | Yes | Yes | Yes |
| Industry effect | Yes | Yes | Yes | Yes | Yes | Yes |
| N Adj. R2 | 12,887 0.170 | 12,887 0.171 | 12,887 0.171 | 12,887 0.170 | 12,887 0.171 | 12,887 0.171 |

Note: ***, ** and * indicate statistical significance at the 1%, 5% and 10% levels, respectively.

Second, we change the control variables, using the share proportion of the five largest shareholders (*conc_top5*) to measure ownership concentration, return on equity (*roe*) to measure financial performance, and operating income (*size_income*) to measure the size of a company. Results are shown in Table 7 from Columns (1) to (3).

Table 7. Robustness checks by changing variable measurements.

| Variables | (1) | (2) | (3) | (4) | (5) | (6) |
|-----------------------------------|------------|------------------------|------------------------|------------|------------------------|------------------------|
| OFDI_d | 2.022 *** | 3.730 *** | 3.078 *** | 2.002 *** | 3.730 *** | 3.078 *** |
| | (0.660) | (0.824) | (0.750) | (0.696) | (0.919) | (0.826) |
| dir_share | | -1.832 *** (0.692) | | | -1.832 *** (0.606) | |
| tmt_share | | | -2.666 *** (0.875) | | | -2.666 *** (0.725) |
| OFDI_d ×dir_share | | -11.183 *** (3.164) | | | -11.183 *** (3.091) | |
| OFDI_d ×tmt_share | | | -14.043 *** (4.475) | | | -14.043 *** (3.537) |
| FDI_d | 1.315 *** | 1.105 ** | 1.134 ** | 1.315 *** | 1.105 ** | 1.134 ** |
| | (0.451) | (0.455) | (0.453) | (0.482) | (0.487) | (0.486) |
| male | 0.812 | 0.669 | 0.787 | 0.812 | 0.669 | 0.787 |
| | (0.521) | (0.522) | (0.521) | (0.491) | (0.492) | (0.490) |
| CEO_age | 0.092 *** | 0.081 *** | 0.085 *** | 0.092 *** | 0.081 *** | 0.085 *** |
| | (0.018) | (0.018) | (0.018) | (0.018) | (0.018) | (0.018) |
| edu | 1.289 *** | 1.151 *** | 1.174 *** | 1.289 *** | 1.151 *** | 1.174 *** |
| | (0.372) | (0.374) | (0.373) | (0.336) | (0.338) | (0.337) |
| forgn_exp | 1.561 *** | 1.593 *** | 1.589 *** | 1.561 *** | 1.593 *** | 1.589 *** |
| | (0.551) | (0.551) | (0.551) | (0.532) | (0.531) | (0.531) |
| conc_top5 | 0.144 *** | 0.148 *** | 0.148 *** | 0.144 *** | 0.148 *** | 0.148 *** |
| | (0.008) | (0.009) | (0.009) | (0.009) | (0.009) | (0.009) |
| size_income | 0.030 *** | 0.029 ** | 0.029 | 0.030 *** | 0.029 ** | 0.029 *** |
| | (0.003) | (0.003) | (0.003) | (0.005) | (0.005) | (0.005) |
| firm_age | 0.082 *** | 0.069 *** | 0.073 *** | 0.082 *** | 0.069 *** | 0.073 *** |
| | (0.025) | (0.026) | (0.025) | (0.024) | (0.025) | (0.025) |
| roa | 0.421 *** | 0.422 *** | 0.421 *** | 0.421 | 0.422 | 0.421 |
| | (0.122) | (0.122) | (0.122) | (0.418) | (0.421) | (0.420) |
| Constant | 8.376 *** | 9.543 *** | 9.106 *** | 8.376 *** | 9.543 *** | 9.106 *** |
| | (1.276) | (1.323) | (1.292) | (1.271) | (1.295) | (1.276) |
| Time effect Province effect | Yes Yes | Yes Yes | Yes Yes | Yes Yes | Yes Yes | Yes Yes |

Sustainability **2022**, 14, 16208 17 of 27

Table 7. Cont.

| Variables | (1) | (2) | (3) | (4) | (5) | (6) |
|--------------------|--------|--------|--------|--------|--------|--------|
| Industry effect | Yes | Yes | Yes | Yes | Yes | Yes |
| N | 13,465 | 13,465 | 13,465 | 13,465 | 13,465 | 13,465 |
| Adj. R2 | 0.196 | 0.197 | 0.197 | 0.196 | 0.197 | 0.197 |

Note: We measure ownership concentration by share proportion of the top 5 largest shareholders (conc_top5), financial performance by return on equity (roe), and the size of a company by operating income (size_income), respectively. *** and ** indicate statistical significance at the 1% and 5% levels, respectively.

Third, the strong assumption of panel data that the samples are serially uncorrelated may be invalid, leading to the underestimation of the standard deviation. Therefore, after correcting the standard deviation of the panel, we perform the regression analysis again. As shown in Columns (4) to (6) of Tables 6 and 7, the models are more robust, and our key conclusions are still valid.

4.3. Heterogeneity Analysis

Is there industry heterogeneity in the impact of OFDI on the sustainable development of enterprises? We further distinguished between the manufacturing and nonmanufacturing samples by keeping control variables and controlling for fixed effects. As shown in Table 8, Columns (1) to (3) report the regression results for the manufacturing sample in Model (1), Model (2) and Model (3) respectively. Columns (4) to (6) report the regression results for the non-manufacturing sample in Model (1), Model (2), and Model (3), respectively. The results suggest that for manufacturing firms, there is a significant positive effect of OFDI on CSR, and equity incentives weaken this positive effect. For nonmanufacturing firms, Column (4) shows that the effect of OFDI on CSR may be insignificant in an average sense. However, after adding the interaction term for regression, the results in Column (5) and Column (6) show that equity incentives similarly weaken the positive effect of OFDI on CSR for non-manufacturing firms. As equity incentives increase, the direction of the effect of OFDI changes from positive to negative. Moreover, according to the interaction term coefficient, shareholding, whether by the board of directors or by the top management team, has a greater debilitating effect on non-manufacturing firms than on manufacturing firms.

To explore the possible regional heterogeneity, the sample was reclassified into eastern, central, and western regions for regression. Columns (1) to (3) of Table 9 report the regression results for the eastern region sample in Model (1), Model (2), and Model (3), respectively. Columns (4) to (6) of Table 9 report the regression results for the central region sample in Model (1), Model (2) and Model (3). Columns (7) to (9) of Table 9 report the results for the western region sample. Although the coefficients on the interaction terms are significant in all regions, there are differences in direction. Equity incentives weaken the positive impact of OFDI on CSR in the eastern and central regions, and this debilitating effect is more pronounced in the central region. Conversely, for companies in the west, equity incentives promote the positive effects of OFDI on CSR. The insignificant coefficients in Column (4) and Column (7) indicate that for the central and western regions, the impact of OFDI on CSR is not identifiable in an average sense. This may indicate that for more developed regions, equity incentive practices are not effective, and companies would do well to find new internal incentives. However, for less developed regions, equity incentives remain a positive means of promoting sustainable business development.

Sustainability **2022**, 14, 16208 18 of 27

Table 8. Regression results with industry heterogeneity.

| | (1) | (2) | (3) | (4) | (5) | (6) | | |
|----------------------|----------------------|----------------------|------------------------|-----------------------|------------------------|------------------------|--|--|
| | Manufacturing | Manufacturing | Manufacturing | Non- Manufacturing | Non- Manufacturing | Non- Manufacturing | | |
| OFDI_d | 1.823 ** (0.754) | 3.205 *** (1.026) | 2.745 *** (0.910) | 1.537 (1.469) | 4.107 ** (1.744) | 3.078 * (1.665) | | |
| dir_share | | 0.267 (0.755) | | | -6.133 *** (1.380) | | | |
| tmt_share | | | -0.996 (0.876) | | | -5.056 *** (1.657) | | |
| OFDI_d ×dir_share | | -8.246 ** (3.338) | | | -21.050 *** (7.259) | | | |
| OFDI_d ×tmt_share | | | -11.334 *** (3.440) | | | -24.238 ** (10.155) | | |
| Control variables | | Yes | Yes | | Yes | Yes | | |
| Constant | 8.453 *** (1.551) | 8.500 *** (1.563) | 8.776 *** (1.551) | 13.778 *** (2.568) | 17.194 *** (2.677) | 15.334 *** (2.603) | | |
| Time effect | Yes | Yes | Yes | Yes | Yes | Yes | | |
| Province effect | Yes | Yes Yes | Yes Yes | Yes Yes | Yes Yes | Yes Yes | | |
| Industry effect N | Yes 9142 | 9142 | 9142 | 4329 | 4329 | 4329 | | |
| Adj. R2 | 0.162 | 0.163 | 0.163 | 0.268 | 0.273 | 0.270 | | |

Note: Standard errors in parentheses; * p < 0.1, ** p < 0.05, *** p < 0.01.

Table 9. Regression results with regional heterogeneity.

| | (1) Eastern | (2) Eastern | (3) Eastern | (4) Central | (5) Central | (6) Central | (7) Western | (8) Western | (9) Western |
|----------------------|----------------------|------------------------|------------------------|-----------------------|------------------------|------------------------|----------------------|----------------------|----------------------|
| OFDI_d | 1.905 ** (0.779) | 3.889 *** (1.047) | 3.086 *** (0.943) | 2.450 (2.035) | 4.836 ** (2.411) | 3.889 * (2.227) | 0.051 (1.974) | -2.078 (2.396) | -1.019 (2.183) |
| dir_share | | -2.068 *** (0.721) | | | 0.753 (2.035) | | | 3.097 (2.279) | |
| tmt_share | | | -2.768 *** (0.847) | | | 0.918 (2.444) | | | 1.036 (2.563) |
| OFDI_d ×dir_share | | -11.767 *** (3.489) | | | -27.504 *** (7.499) | | | 18.689 ** (7.744) | |
| OFDI_d ×tmt_share | | | -14.015 *** (3.886) | | | -33.455 *** (7.981) | | | 23.367 * (13.988) |
| Control variables | Yes | Yes | Yes | Yes | Yes | Yes | Yes | Yes | Yes |
| Constant | 7.777 *** (1.554) | 9.354 *** (1.580) | 8.718 *** (1.559) | 16.976 *** (3.565) | 16.887 *** (3.805) | 16.694 *** (3.667) | 10.997 ** (4.571) | 8.480 * (4.757) | 10.440 ** (4.612) |
| Time effect | Yes | Yes | Yes | Yes | Yes | Yes | Yes | Yes | Yes |
| Province effect | Yes | Yes | Yes | Yes | Yes | Yes | Yes | Yes | Yes |
| Industry effect | Yes | Yes | Yes | Yes | Yes | Yes | Yes | Yes | Yes |
| N Adj. R2 | 10,076 0.202 | 10,076 0.203 | 10,076 0.203 | 1899 0.244 | 1899 0.246 | 1899 0.245 | 1491 0.243 | 1491 0.244 | 1491 0.242 |

Note: Standard errors in parentheses; * p < 0.1, ** p < 0.05, *** p < 0.01.

5. Discussion

All the hypotheses proposed in this paper are accepted. First, this study confirms the existence of reverse spillover effects of OFDI in China, which is consistent with the studies [137,138]. The innovation is that this paper examines multiple aspects of CSR using a combination of indicators, rather than considering only green productivity. This is an important addition to the existing literature, as there is a growing body of practical

Sustainability **2022**, 14, 16208 19 of 27

experience that proves that in addition to productive capacity, employee and social welfare are also important components in promoting corporate sustainable development. Notably, similar to much of the literature, we use a one-period lagged OFDI, which preserves the impact of OFDI on CSR. At the same time, it can avoid some endogeneity issues. On the other hand, since the effect of OFDI on CSR is lagged [139], using lagged OFDI also makes the empirical evidence more explanatory.

Furthermore, Zhou et al. [66] find that financial development and human capital weakened the positive relationship between OFDI and domestic innovation performance in developed countries. This study also reached a similar conclusion in the Chinese market that there is a weakening effect of equity incentives on the positive relationship between OFDI and CSR. However, our research is more focused on corporate micro-governance structures. We conjecture that the disincentive effect of equity incentives is more pronounced in mature markets and that equity incentives are also negative in the Chinese market as China is going through an important phase of economic transition. At the same time, the study's findings indicate that equity incentives may not be as effective as commonly believed. As other researchers have worried, the fragmentation of corporate control reduces the value of the business [46]. If directors or TMT hold a significant amount of equity in a company, they would have enough voting power and influence to ensure that they receive higher remuneration, resulting in them not being focused on profit maximization goals. Moreover, the more shares management holds, the less likely the company is to be acquired, which also makes shareholder pressure on management weaker. Of course, this phenomenon may also be attributed to various reasons such as the immaturity of China's equity incentive system and the insufficient degree of equity incentive.

Finally, the existence of heterogeneity in OFDI spillover effects is well documented [133,140]. We build on this by further discussing the regional heterogeneity and industry heterogeneity of the moderating effect of equity incentives, providing a clearer dissection of the impact mechanism. Equity incentives show a facilitating effect in the more economically backward western region, as opposed to the economically developed east-central region. This finding further supports the conjecture above that the role of equity incentives is dissipating in more mature markets.

6. Conclusions

Based on the evidence from China, this study proves that corporate globalization is an effective channel for improving corporate ecological management reflected by the positive impact of OFDI on corporate sustainable development performance.

This paper uses CSR scores for Chinese listed companies provided by the Hexun Database to measure corporate sustainability on multiple dimensions rather than one-sided. On this basis, this study innovatively includes equity incentives as a moderator and subdivides them into shares held by the board of directors and shares held by the TMT. A three-way fixed-effects empirical model is used, and interaction terms are added to examine the moderating effect of equity incentives. The empirical results reveal that OFDI has a significant positive effect on CSR, while equity incentives weaken this positive effect, and a subsequent robustness test confirms this conclusion. From the perspective of industry heterogeneity, the disincentive effect of equity incentives is more pronounced in the non-manufacturing sector. From the perspective of regional heterogeneity, equity incentives show a disincentive effect in the eastern and central regions, while they show a facilitating effect in the western region.

6.1. Implications

The findings of this study have various academic and management implications. First, the study contributes to a theoretical understanding of the relationship between OFDI, corporate sustainability, and the internal incentive structure of firms. The incentive approach determines the direction of the manager's efforts, and the manager is the main player in deciding and implementing the OFDI strategy, while OFDI has a significant

Sustainability **2022**, 14, 16208 20 of 27

contribution to CSR. Thus, the incentive approach assumes a moderating role in this process. Our research reaffirms the existence of reverse spillover effects [137,138], particularly for emerging markets. Meanwhile, the empirical results provide preliminary evidence of the disincentive effect of equity incentives, which fills a gap in the literature related to the way incentives are viewed from an OFDI perspective. This finding also refutes the view put forward in some literature that equity incentives are efficient [141].

Our research may have broader implications in management practice. The expansion of Chinese OFDI is part of China's rise as a major economy, indicating its increasing integration with the world economy. While much has been discussed about the impact of this macro development on CSR in Chinese firms, little is known about the micro governance of the firms involved. This paper argues that the latter should not be ignored, as corporate governance issues are an important consideration for Chinese companies going global. They must improve their governance levels and structures to adapt to global rules. Exploring what role corporate governance plays in the impact of OFDI on CSR is an important guide for firms' external strategies. Moreover, the study finds that the actual effects of equity incentives are contrary to the original intention and are likely to be more pronounced in mature markets. This suggests that for the Chinese market, which is undergoing an economic transition, a change in incentives is an important element to ensure sustainable business development. Improving the shareholding structure and diversifying incentives is an urgent necessity for Chinese companies trying to go global.

6.2. Recommendations

In a practical sense, we provide some suggestions for CSR investment, corporate internal governance, and human resource management. Companies can better improve the ability to conduct corporate sustainable development and explore new practices of sustainability by combining globalization strategies with their internal governance.

First, managers of emerging market corporations should take a proactive approach to develop global business and expand investment abroad. In the context of OFDI, the knowledge transfer process can be accumulated in the course of foreign exchange and learning by enterprises, thus promoting sustainable development and corporate social responsibility. On the one hand, corporate sustainability is more about a kind of consciousness [142]. Companies have more forward-looking beliefs in long-term development planning [143]. As companies implementing OFDI become more open to global trends in CSR management solutions, the more OFDI impacts the international environment, and the more likely they are to spur themselves on with corporate sustainability standards [86]. On the other hand, corporate sustainability may also be more than an awareness or image but covers more substantial elements. Active participation in global value chains is an effective way for MNEs to gain innovation and management capabilities, which help them gain sustainable momentum in terms of technology and governance. Hence, decision-makers in companies should actively promote global policies to develop the advantages of investment and thus enhance core competencies. After enough accumulation of advanced corporate sustainable development knowledge during OFDI and improving their ecological management, MNEs may be able to further enhance their competitive position in the international arena [144].

Second, we suggest that policies for encouraging the flow of TMT should be established. The openness of CEOs is of great significance, especially the overseas working experience. It is undeniable that overseas work experience, especially in developed regions, often brings CEOs a broader vision and advanced management knowledge. Additionally, CEOs with overseas work experience have certain information and resource advantages that may be more conducive to the grounding of corporate decisions. Those advantages can improve corporate governance and let them focus more on long-term corporate investment decisions [145]. Sater and Dixon-Fowler prove that CEOs with overseas experience are more inclined to focus on the interests of stakeholders and on the CSR performance of the company [146]. Therefore, encouraging TMT mobility and absorbing more open and

Sustainability **2022**, 14, 16208 21 of 27

progressive managers to bring fresh blood into the management team is conducive to opening up new perspectives for the company's long-term development.

Third, in order to balance corporate's financial goals, other incentives such as health and safety, honor and knowledge sharing should be utilized. The agency problem in the company's internal governance is likely to derail the role of equity incentives from the capital track. As a common incentive method, equity distribution does not have beneficial effects on companies' performance of social responsibility during the process of corporate globalization. Particularly in more developed regions, equity incentives have instead failed to have a positive effect, which should be taken seriously. Hence, through the course of improving corporate sustainable development performance, other incentives than equity should be used. For instance, Derchi et al. find that the compensation contracts of named executive officers linked to CSR can facilitate the performance of CSR [147]. This is informative for effective incentives to promote CSR performance. Companies can also adopt a more systematic balanced scorecard approach to executive compensation [148], which significantly shifts the focus from short-term financial goals to long-term value creation and creates more room for corporate social responsibility.

Finally, we recommend that policy makers continue to improve the corporate share-holding system. Considering the significant disincentive moderation effect of equity incentives in economically developed regions, we believe that the improvement of laws and regulations has not kept pace with economic development, resulting in equity incentives becoming a tool for managers to seek private gain. For example, in the case concluded by the CSRC on 25 May 2018, ESSAT executives used their effective control to manipulate the share price of the company to gain substantial profits [149]. As equity incentives are still the motivational tool used by many listed companies, it is difficult to create another incentive to replace it entirely. Therefore, thinking about how to improve the relevant regulation may be an important way to reverse the bad trend of share incentives. At present, the reform of the shareholding system has become an important issue for Chinese companies to solve. The relevant market needs to further clarify the regulatory body, determine shareholding powers and improve the employee equity exit mechanism.

6.3. Limitations

This paper also has some limitations. In view of data availability, we verify the spillover effects of OFDI on corporate sustainable development performance without incorporating specific channels through which OFDI affects corporate sustainable development performance into our models. On the other hand, the characteristics of host countries have not been considered. Given corporate sustainable development attracting attention globally, it is necessary to conduct international comparative research and investigate the differences in OFDI spillover efficiency on corporate sustainable development performance between developed and developing countries. In addition, this paper also needs to have more qualitative research. We selected data such as the shareholdings of directors and TMT from CSMAR. These data are not problematic. However, on the one hand they may have lags, and on the other hand, there may be problems with strategic disclosure, thus affecting the results of the study. We can follow up with further qualitative research, such as interview studies for firms with characteristics.

Author Contributions: Conceptualization, B.F., C.S. and X.Z.; methodology, B.F., C.S. and X.Z.; software, B.F., C.S. and X.Z.; validation, T.S., X.L., C.S. and X.Z.; formal analysis, T.S., B.F., X.L., C.S. and X.Z.; investigation, T.S., X.L., C.S. and X.Z.; resources, C.S. and X.Z.; data curation, B.F.; writing—original draft preparation, T.S., B.F., X.L., C.S. and X.Z.; writing—review and editing, T.S., X.L., X.S., C.S. and X.Z.; visualization, T.S. and X.L.; supervision, C.S. and X.Z.; project administration, C.S. and X.Z.; funding acquisition, C.S. and X.Z. All authors have read and agreed to the published version of the manuscript.

Funding: This research was funded by the National Natural Science Foundation of China, grant number 72003067; the Shanghai Municipal Foundation for Philosophy and Social Science, grant number 2018ECK007; the Pujiang Talent Program, grant number 18PJC033; and the Fundamental

Sustainability **2022**, 14, 16208 22 of 27

Research Funds for the Central Universities, grant numbers 2021QKT009, 2018ECNU-HLYT005, 2021ECNU-YYJ024.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: The data presented in this study are available on request from the corresponding author.

Conflicts of Interest: The authors declare no conflict of interest.

Appendix A

Table A1. Correlation matrix.

| Variables | (1) | (2) | (3) | (4) | (5) | (6) | (7) | (8) | (9) | (10) | (11) | (12) | (13) | (14) | (15) | (16) |
|------------------|---------|---------|---------|---------|---------|---------|---------|--------|---------|---------|---------|---------|---------|---------|--------|--------|
| (1) CSR | 1.0000 | | | | | | | | | | | | | | | |
| (2) OFDI_d | 0.0301 | 1.0000 | | | | | | | | | | | | | | |
| (3) FDI_d | 0.1379 | 0.0219 | 1.0000 | | | | | | | | | | | | | |
| (4) dir_share | -0.0696 | 0.0038 | -0.1360 | 1.0000 | | | | | | | | | | | | |
| (5) tmt_share | -0.0564 | -0.0062 | -0.1010 | 0.7733 | 1.0000 | | | | | | | | | | | |
| (6) male | 0.0147 | -0.0077 | 0.0161 | -0.0652 | -0.0246 | 1.0000 | | | | | | | | | | |
| (7) CEO_age | 0.0145 | -0.0244 | -0.0174 | -0.2903 | -0.2125 | 0.0003 | 1.0000 | | | | | | | | | |
| (8) edu | 0.0558 | 0.0020 | 0.0304 | -0.1148 | -0.0899 | 0.0523 | 0.0085 | 1.0000 | | | | | | | | |
| (9) forgn_exp | 0.0023 | 0.0285 | 0.0226 | 0.0445 | 0.0464 | -0.0112 | -0.0062 | 0.0445 | 1.0000 | | | | | | | |
| (10) conc_top3 | 0.1718 | 0.0154 | 0.1721 | 0.0999 | 0.0939 | -0.0473 | -0.0587 | 0.0079 | 0.0249 | 1.0000 | | | | | | |
| (11) conc_top5 | 0.1726 | 0.0152 | 0.1685 | 0.1789 | 0.1528 | -0.0493 | -0.1007 | 0.0025 | 0.0334 | 0.9732 | 1.0000 | | | | | |
| (12) size_assets | 0.1121 | -0.0035 | 0.1482 | -0.0636 | -0.0479 | 0.0188 | 0.0837 | 0.0321 | -0.0091 | 0.1408 | 0.1347 | 1.0000 | | | | |
| (13) size_income | 0.1065 | -0.0026 | 0.1483 | -0.0777 | -0.0608 | 0.0196 | 0.1029 | 0.0377 | 0.0150 | 0.1629 | 0.1503 | 0.3770 | 1.0000 | | | |
| (14) firm_age | -0.0301 | -0.0184 | -0.0840 | -0.2489 | -0.1889 | 0.0085 | 0.3047 | 0.0143 | -0.0030 | -0.1545 | -0.1784 | 0.0710 | 0.0134 | 1.0000 | | |
| (15) roe | 0.0107 | -0.0003 | 0.0206 | -0.0017 | -0.0012 | 0.0089 | -0.0131 | 0.0065 | -0.0007 | -0.0023 | -0.0022 | 0.0006 | 0.0002 | 0.0035 | 1.0000 | |
| (16) roa | 0.0344 | 0.0005 | 0.0060 | 0.0080 | 0.0078 | 0.0006 | -0.0072 | 0.0083 | 0.0013 | 0.0077 | 0.0107 | -0.0032 | -0.0008 | -0.0042 | 0.2996 | 1.0000 |

References

- 1. Zanten, J.A.V.; Tulder, R.V. Improving companies' impacts on sustainable development: A nexus approach to the SDGS. *Bus. Strateg. Environ.* **2021**, *30*, 3703–3720. [CrossRef]
- 2. Ding, X.; Ye, L.; Yang, Y.; Efimova, O.; Steblyanskaya, A.; Zhang, J. The Impact Mechanism of Environmental Information Disclosure on Corporate Sustainability Performance—Micro-Evidence from China. *Sustainability* **2022**, *14*, 12366. [CrossRef]
- 3. Tien, N.H.; Anh, D.B.H.; Ngoc, N.M. Corporate financial performance due to sustainable development in Vietnam. *Corp. Soc. Responsib. Environ. Manag.* **2020**, *27*, 694–705. [CrossRef]
- 4. The KPMG Survey of Corporate Responsibility Reporting 2017. Available online: Kpmg.com/crreporting (accessed on 9 October 2022).
- 5. Li, J.; Wu, X. The effect of CSR on acquirer returns of cross-border M & As in an emerging market–a bane or a boom? *Appl. Econ. Lett.* **2022**, *29*, 1–6. [CrossRef]
- 6. Shen, Y.; Wu, J.; Wu, S. City-chief turnover and place—Based policy change: Evidence from China. *J. Reg. Sci.* **2022**, *62*, 1296–1328. [CrossRef]
- 7. Yin, J.; Jamali, D. Strategic corporate social responsibility of multinational companies subsidiaries in emerging markets: Evidence from China. *Long Range Plan.* **2016**, *49*, 541–558. [CrossRef]
- 8. Park, J.K.; Lee, D.J.; Lee, K. The Determinants of foreign direct investment in R & D: Different inducement effects of private and public R&D in developed and developing host countries. *Singap. Econ. Rev.* **2022**, *67*, 923–951. [CrossRef]
- 9. Peng, D.; Yu, J.; Kong, Q. OFDI and firms' sustainable productive capacity: Evidence from Chinese industrial firms. *Int. Rev. Econ. Financ.* **2022**, *83*, 641–652. [CrossRef]
- 10. Zhang, L. The knowledge spillover effects of FDI on the productivity and efficiency of research activities in China. *China Econ. Rev.* **2017**, *42*, 1–14. [CrossRef]
- 11. Hu, F. Technology seeking, R & D localization and comparative advantage: Evidence from Chinese R & D-oriented OFDI. *Appl. Econ. Lett.* **2022**, *29*, 1–6. [CrossRef]
- 12. Wang, M.L.; Pang, S.L.; Wang, F.; Guo, X.; He, Z.X. Dynamic interaction between outward foreign direct investment and home country industrial upgrading: Regional differences in China. *Growth Chang.* **2021**, *52*, 2293–2317. [CrossRef]
- 13. Peng, C.; Jia, X.; Tang, Y. Does OFDI promote high-quality development of enterprises? Evidence from China. *Am. J. Ind. Bus. Manag.* **2022**, *12*, 153–179. [CrossRef]
- 14. Lyles, M.; Li, D.; Yan, H. Chinese outward foreign direct investment performance: The role of learning. *Manag. Organ. Res.* **2014**, 10, 411–437. [CrossRef]
- 15. Wong, Z.; Chen, A.; Peng, D.; Kong, Q. Does technology-seeking OFDI improve the productivity of Chinese firms under the COVID-19 pandemic? *Glob. Financ. J.* **2022**, *51*, 100675. [CrossRef]
- 16. Smarzynska Javorcik, B. Does foreign direct investment increase the productivity of domestic firms? In search of spillovers through backward linkages. *Am. Econ. Rev.* **2004**, *94*, 605–627. [CrossRef]

Sustainability **2022**, 14, 16208 23 of 27

17. Pananond, P. Where do we go from here?: Globalizing subsidiaries moving up the value chain. *J. Int. Manag.* **2013**, *19*, 207–219. [CrossRef]

- 18. Jiang, M.; Luo, S.; Zhou, G. Financial development, OFDI spillovers and upgrading of industrial structure. *Technol. Forecast. Soc. Chang.* **2020**, *155*, 119974. [CrossRef]
- 19. Shi, J. Formal Institutional Distance and Innovation from OFDI: Evidence from China. Sustainability 2022, 14, 5368. [CrossRef]
- 20. Long, W.; Luo, L.; Sun, H.; Zhong, Q. Does going abroad lead to going green? Firm outward foreign direct investment and domestic environmental performance. *Bus. Strateg. Environ.* **2022**, *31*, 1–15. [CrossRef]
- 21. Peng, H.; Yu, J. Absorptive capacity and quality upgrading effect of OFDI: Evidence from China. *Pac. Econ. Rev.* **2021**, *26*, 651–671. [CrossRef]
- 22. Chen, T.G.; Lin, G.; Yabe, M. Impact of outward FDI on firms' productivity over the food industry: Evidence from China. *China Agric. Econ. Rev.* **2019**, *11*, 655–671. [CrossRef]
- 23. Song, Y.; Hao, F.; Hao, X.; Gozgor, G. Economic policy uncertainty, outward foreign direct investments, and green total factor productivity: Evidence from firm-level data in China. *Sustainability* **2021**, *13*, 2339. [CrossRef]
- 24. Huang, Y.; Zhang, Y. How does outward foreign direct investment enhance firm productivity? A heterogeneous empirical analysis from Chinese manufacturing. *China Econ. Rev.* **2017**, *44*, 1–15. [CrossRef]
- 25. Huang, Y.; Zhong, C.; Chen, H. The combined effect of foreign direct investment on firm productivity. *Ekon. Istraz.* **2022**, *35*, 1–18. [CrossRef]
- 26. Sahasranamam, S.; Arya, B.; Mukundhan, K.V. Dual institutional embeddedness and home country CSR Engagement: Evidence from Indian MNEs. *J. Bus. Resh.* **2022**, *141*, 163–174. [CrossRef]
- 27. Sahasranamam, S.; Nandakumar, M.K. Individual capital and social entrepreneurship: Role of formal institutions. *J. Bus. Res.* **2020**, *107*, 104–117. [CrossRef]
- 28. Moneva, J.M.; Ortas, E. Are stock markets influenced by sustainability matter? Evidence from European companies. *Int. J. Sustain. Econ.* **2008**, *1*, 1–16. [CrossRef]
- 29. Dong, X.; Yu, C.; Hwang, Y.S. The Effects of Reverse Knowledge Spillover on China's Sustainable Development: Sustainable Development Indicators Based on Institutional Quality. *Sustainability* **2021**, *13*, 1628. [CrossRef]
- 30. Zhang, Q.; Naqvi, S.A.A.; Shah, S.A.R. The Contribution of Outward Foreign Direct Investment, Human Well-Being, and Technology toward a Sustainable Environment. *Sustainability* **2021**, *13*, 11430. [CrossRef]
- 31. Ren, S.; Hao, Y.; Wu, H. The role of outward foreign direct investment (OFDI) on green total factor energy efficiency: Does institutional quality matters? Evidence from China. *Res. Policy* **2022**, *76*, 102587. [CrossRef]
- 32. Yang, T.; Dong, Q.; Du, Q.; Du, M.; Dong, R.; Chen, M. Carbon dioxide emissions and Chinese OFDI: From the perspective of carbon neutrality targets and environmental management of home country. *J. Environ. Manag.* **2021**, 295, 113120. [CrossRef] [PubMed]
- 33. Naciti, V. Corporate governance and board of directors: The effect of a board composition on firm sustainability performance. *J. Clean. Prod.* **2019**, 237, 117727. [CrossRef]
- 34. Orazalin, N.; Baydauletov, M. Corporate social responsibility strategy and corporate environmental and social performance: The moderating role of board gender diversity. *Corp. Soc. Responsib. Environ. Manag.* **2020**, 27, 1664–1676. [CrossRef]
- 35. Rao, K.; Tilt, C. Board composition and corporate social responsibility: The role of diversity, gender, strategy and decision making. *J. Bus. Ethics* **2016**, 138, 327–347. [CrossRef]
- 36. King, T.; Srivastav, A.; Williams, J. What's in an education? Implications of CEO education for bank performance. *J. Corp. Financ.* **2016**, *37*, 287–308. [CrossRef]
- 37. Syakhroza, A.; Diyanty, V.; Dewo, S.A. Top management team (TMT) age diversity and firm performance: The moderating role of the effectiveness of TMT meetings. *Team Perform. Manag.* **2021**, 27, 486–503. [CrossRef]
- 38. Zhang, Y.; Ayoko, O.B.; Liang, Q. The joint influence of CEO succession types and CEO-TMT faultline on firm's strategic change. *J. Bus. Res.* **2021**, *126*, 137–152. [CrossRef]
- 39. Aboody, D.; Kasznik, R. Discussion of executive stock-based compensation and firms' cash payout: The role of shareholders' tax-related payout preferences. *Rev. Account. Stud.* **2008**, *13*, 216–251. [CrossRef]
- 40. Lamberti, L.; Lettieri, E. CSR practices and corporate strategy: Evidence from a longitudinal case study. *J. Bus. Ethics* **2009**, *87*, 153–168. [CrossRef]
- 41. Crisóstomo, V.L.; Freire, F.; Vasconcellos, F.C.D. Corporate social responsibility, firm value and financial performance in Brazil. *Soc. Responsib. J.* **2011**, *7*, 295–309. [CrossRef]
- 42. Eisenhardt, K.M. Agency theory: An assessment and review. Acad. Manag. Rev. 1989, 14, 57–74. [CrossRef]
- 43. Akpan, E.O.; Amran, N.A. Board characteristics and company performance: Evidence from Nigeria. *J. Financ. Account.* **2014**, 2, 81–89. [CrossRef]
- 44. Eng, L.L.; Mak, Y.T. Corporate governance and voluntary disclosure. J. Account. Public Policy 2003, 22, 325–345. [CrossRef]
- 45. Hidalgo, R.L.; García-Meca, E.; Martínez, I. Corporate governance and intellectual capital disclosure. *J. Bus. Ethics* **2013**, *100*, 483–495. [CrossRef]
- 46. Fama, E.F.; Jensen, M.C. Separation of ownership and control. J. Law Econ. 1983, 26, 301–325. [CrossRef]
- 47. Antounian, C.; Dah, M.A.; Harakeh, M. Excessive managerial entrenchment, corporate governance, and firm performance. *Res. Int. Bus. Financ.* **2021**, *56*, 101392. [CrossRef]

Sustainability **2022**, 14, 16208 24 of 27

- 48. Collins, D.; Huang, H. Management entrenchment and the cost of equity capital. J. Bus. Res. 2011, 64, 356–362. [CrossRef]
- 49. Fallah Shayan, N.; Mohabbati-Kalejahi, N.; Alavi, S.; Zahed, M.A. Sustainable development goals (SDGs) as a framework for corporate social responsibility (CSR). *Sustainability* **2022**, *14*, 1222. [CrossRef]
- 50. McWilliams, A.; Siegel, D.S. Creating and capturing value: Strategic corporate social responsibility, resource-based theory, and sustainable competitive advantage. *J. Manag.* **2011**, *37*, 1480–1495. [CrossRef]
- 51. Baumgartner, R.J. Managing corporate sustainability and CSR: A conceptual framework combining values, strategies and instruments contributing to sustainable development. *Corp. Soc. Responsib. Environ. Manag.* **2014**, *21*, 258–271. [CrossRef]
- 52. Poddar, A.; Narula, S.A.; Zutshi, A. A study of corporate social responsibility practices of the top Bombay Stock Exchange 500 companies in India and their alignment with the Sustainable Development Goals. *Corp. Soc. Responsib. Environ. Manag.* **2019**, 26, 1184–1205. [CrossRef]
- 53. Duque-Grisales, E.; Aguilera-Caracuel, J.; Guerrero-Villegas, J.; García-Sánchez, E. Can a proactive environmental strategy improve Multilatinas' level of internationalization? The moderating role of board independence. *Bus. Strateg. Environ.* **2020**, 29, 291–305. [CrossRef]
- 54. Fukuda, K.; Ouchida, Y. Corporate social responsibility (CSR) and the environment: Does CSR increase emissions? *Energy Econ.* **2020**, *92*, 104933. [CrossRef]
- 55. Xie, X.; Jia, Y.; Meng, X.; Li, C. Corporate social responsibility, customer satisfaction, and financial performance: The moderating effect of the institutional environment in two transition economies. *J. Clean. Prod.* **2017**, *150*, 26–39. [CrossRef]
- 56. Shen, C.; Fang, B.; Zhou, X. The Relationship between Corporate Sustainable Development Performance, Investor Sentiment, and Managerial Overconfidence. *Sustainability* **2022**, *14*, 10606. [CrossRef]
- 57. Joseph, E. Promoting corporate social responsibility: Is market-based regulation sufficient? *IPPR Progress. Rev.* **2002**, *9*, 96–101. [CrossRef]
- 58. Gond, J.P.; Kang, N.; Moon, J. The government of self-regulation: On the comparative dynamics of corporate social responsibility. *Econ. Soc.* **2011**, 40, 640–671. [CrossRef]
- 59. Singhapakdi, A.; Lee, D.J.; Sirgy, M.J.; Senasu, K. The impact of incongruity between an organization's CSR orientation and its employees' CSR orientation on employees' quality of work life. *J. Bus. Res.* **2015**, *68*, 60–66. [CrossRef]
- 60. Golob, U.; Podnar, K. Corporate marketing and the role of internal CSR in employees' life satisfaction: Exploring the relationship between work and non-work domains. *J. Bus. Res.* **2021**, *131*, 664–672. [CrossRef]
- 61. Tien, N.H.; Hung Anh, D.B. Gaining comparative advantage from CSR policy change—Cases of international corporations in Vietnam. *Pol. J. Manag. Stud.* **2018**, *18*, 403–417. [CrossRef]
- 62. Hong, J.; Zhou, C.; Wu, Y.; Wang, R.; Marinova, D. Technology gap, reverse technology spillover and domestic innovation performance in outward foreign direct investment: Evidence from China. *China World Econ.* **2019**, 27, 1–23. [CrossRef]
- 63. Chen, V.Z.; Li, J.; Shapiro, D.M. International reverse spillover effects on parent firms: Evidences from emerging market MNEs in developed markets. *Eur. Manag. J.* **2012**, *30*, 204–218. [CrossRef]
- 64. Dunlap, D.; McDonough, E.F., III; Mudambi, R.; Swift, T. Making up is hard to do: Knowledge acquisition strategies and the nature of new product innovation. *J. Prod. Innov. Manag.* **2016**, *33*, 472–491. [CrossRef]
- 65. Katila, R.; Ahuja, G. Something old, something new: A longitudinal study of search behavior and new product introduction. *Acad. Manag. Ann.* **2020**, *45*, 1183–1194. [CrossRef]
- 66. Zhou, C.; Hong, J.; Wu, Y.; Marinova, D. Outward foreign direct investment and domestic innovation performance: Evidence from China. Technol. *Anal. Strateg. Manag.* **2019**, *31*, 91–95. [CrossRef]
- 67. Zhou, Y.; Jiang, J.; Ye, B.; Hou, B. Green spillovers of outward foreign direct investment on home countries: Evidence from China's province-level data. *J. Clean. Prod.* **2019**, 215, 829–844. [CrossRef]
- 68. Chen, K.M.; Yang, S.F. Impact of Outward Foreign Direct Investment on Domestic R&D Activity: Evidence from Taiwan's Multinational Enterprises in Low-wage Countries. *Asian Econ. J.* **2013**, 27, 17–38. [CrossRef]
- 69. Banerjee, S.; Prahbu, J.C.; Chandy, R.K. Indirect learning: How emerging-market firms grow in developed markets. *J. Mark.* **2015**, 79, 10–28. [CrossRef]
- 70. Lu, J.; Liu, X.; Wang, H. Motives for outward FDI of Chinese private firms firm resources, industry dynamics, and government policies. *Manag. Organ. Rev.* **2011**, *7*, 223–248. [CrossRef]
- 71. Luo, Y.; Tung, R.L. International expansion of emerging market enterprises: A springboard perspective. *J. Int. Bus. Stud.* **2007**, *38*, 481–498. [CrossRef]
- 72. Child, J.; Rodrigues, S.B. The internationalization of Chinese firms: A case for theoretical extension? *Manag. Organ. Rev.* **2005**, *1*, 381–410. [CrossRef]
- 73. Bhaumik, S.; Driffield, N.; Gaur, A.; Mickiewicz, T.; Vaaler, P. Corporate governance and MNE strategies in emerging economies. *J. World Bus.* **2019**, *54*, 234–243. [CrossRef]
- 74. Khan, M.A.; Ali, S.T.; Yang, Z.; Ali, F.; Sarwar, Z. Outward foreign direct investment and corporate green innovation: An institutional pressure perspective. S. Afr. J. Bus. Manag. 2020, 51, 1–12.
- 75. Lazarides, T.; Drimpetas, E. Corporate governance regulatory convergence: A remedy for the wrong problem. *Int. J. Law Manag.* **2010**, *52*, 182–192. [CrossRef]
- 76. Berrone, P.; Fosfuri, A.; Gelabert, L.; Gomez-Mejia, L.R. Necessity as the mother of 'green' inventions: Institutional pressures and environmental innovations. *Strateg. Manag. J.* **2013**, *34*, 891–909. [CrossRef]

Sustainability **2022**, 14, 16208 25 of 27

77. Klewitz, J.; Hansen, E.G. Sustainability-oriented innovation of SMEs: A systematic review. *J. Clean. Prod.* **2014**, *65*, 57–75. [CrossRef]

- 78. Hao, Y.; Guo, Y.; Guo, Y.; Wu, H.; Ren, S. Does outward foreign direct investment (OFDI) affect the home country's environmental quality? The case of China. *Struct. Chang. Econ. Dyn.* **2020**, *52*, 109–119. [CrossRef]
- 79. Gihleb, R.; Giuntella, O.; Stella, L.; Wang, T. Industrial robots, workers' safety, and health. Labour Econ. 2022, 78, 102205. [CrossRef]
- 80. McLean, G.; Osei-Frimpong, K.; Barhorst, J. Alexa, do voice assistants influence consumer brand engagement?-Examining the role of AI powered voice assistants in influencing consumer brand engagement. *J. Bus. Res.* **2021**, 124, 312–328. [CrossRef]
- 81. Shahzad, M.; Qu, Y.; Zafar, A.U.; Rehman, S.U.; Islam, T. Exploring the influence of knowledge management process on corporate sustainable performance through green innovation. *J. Knowl. Manag.* **2020**, *24*, 2079–2106. [CrossRef]
- 82. Charumathi, B.; Padmaja, G. The impact of regulations and technology on corporate social responsibility disclosures—Evidence from Maharatna Central public sector enterprises in India. Australas. *Account. Bus. Financ. J.* **2018**, *12*, 5–28. [CrossRef]
- 83. Huang, X.S.; Sun, L. Managerial ability and real earnings management. Adv. Account. 2017, 39, 91–104. [CrossRef]
- 84. Salehi, M.; Shiri, M.M.; Hossini, S.Z. 2019 The relationship between managerial ability, earnings management and internal control quality on audit fees in Iran. *Int. J. Product. Perform. Manag.* **2019**, *69*, *685–703*. [CrossRef]
- 85. Oskouei, Z.H.; Sureshjani, Z.H. Studying the relationship between managerial ability and real earnings management in economic and financial crisis conditions. *Int. J. Financ. Econ.* **2021**, 26, 4574–4589. [CrossRef]
- 86. Wang, H.; Hu, X. China's "going-out" strategy and corporate social responsibility: Preliminary evidence of a "boomerang effect". *J. Contemp. China* **2017**, *26*, 820–833. [CrossRef]
- 87. Shin, H.D.; Lee, N.R.; Park, J.H. Differential effects of strong corporate governance on both professional and voluntary corporate social responsibility activities of the firm. *Emerg. Mark. Financ. Trade* **2015**, *51*, S2–S10. [CrossRef]
- 88. Chan, M.C.; Watson, J.; Woodliff, D. Corporate governance quality and CSR disclosures. J. Bus. Ethics 2014, 125, 59–73. [CrossRef]
- 89. Boisot, M.; Meyer, M.W. Which way through the open door? Reflections on the internationalization of Chinese firms. *Manag. Organ. Rev.* **2008**, *4*, 349–365. [CrossRef]
- 90. Li, X.; Zhou, W.; Hou, J. Research on the impact of OFDI on the home country's global value chain upgrading. *Int. Rev. Financ. Anal.* **2021**, *77*, 101862. [CrossRef]
- 91. Kong, Q.; Guo, R.; Wang, Y.; Sui, X.; Zhou, S. Home-country environment and firms' outward foreign direct investment decision: Evidence from Chinese firms. *Econ. Model.* **2020**, *85*, 390–399. [CrossRef]
- 92. Knoerich, J. How does outward foreign direct investment contribute to economic development in less advanced home countries? *Oxf. Dev. Stud.* **2017**, *45*, 443–459. [CrossRef]
- 93. Cabeza-García, L.; Fernández-Gago, R.; Nieto, M. Do board gender diversity and director typology impact CSR reporting? *Eur. Manag. Rev.* **2017**, *15*, 559–575. [CrossRef]
- 94. Saeed, A.; Riaz, H.; Liedong, T.A.; Rajwani, T. The impact of TMT gender diversity on corporate environmental strategy in emerging economies. *J. Bus. Res.* **2022**, *141*, 536–551. [CrossRef]
- 95. Al-Shammari, M.; Rasheed, A.; Al-Shammari, H.A. CEO narcissism and corporate social responsibility: Does CEO narcissism affect CSR focus? *J. Bus. Res.* **2019**, *104*, 106–117. [CrossRef]
- 96. Huber, R.; Hirsch, B. Behavioral effects of sustainability—Oriented incentive systems. *Bus. Strateg. Environ.* **2017**, 26, 163–181. [CrossRef]
- 97. Cai, W.; Lee, E.; Wu, Z.; Xu, A.L.; Zeng, C.C. Do economic incentives of controlling shareholders influence corporate social responsibility disclosure? A natural experiment. *Int. J. Account.* **2017**, *52*, 238–250. [CrossRef]
- 98. Bian, C.; Gan, C.; Li, Z.; Hu, B. Corporate social responsibility engagement, corporate financial performance and CEO characteristics. *Int. J. Bus. Gov. Ethics* **2016**, *11*, 243–265. [CrossRef]
- 99. Jensen, M.C.; Meckling, W.H. Theory of the firm: Managerial behaviour, agency costs and ownership structure. *J. Financ. Econ.* **1976**, *3*, 305–360. [CrossRef]
- 100. Sun, Y.; Xia, J. Stakeholder interest to mitigate the agency problem in enterprise innovation and the moderating effect of ownership concentration and financial constraints. *Creativity Innov. Manag.* **2022**, *31*, 599–613. [CrossRef]
- 101. Wu, J.; Tu, R. CEO stock option pay and R & D spending: A behavioral agency explanation. *J. Bus. Res.* **2007**, *60*, 482–492. [CrossRef]
- 102. Lerner, J.; Wulf, J. Innovation and incentives: Evidence from corporate R & D. Rev Econ Stat. 2007, 89, 634-644. [CrossRef]
- 103. Cheng, Q.; Warfield, T.D. Equity incentives and earnings management. Account. Rev. 2005, 80, 441–476. [CrossRef]
- 104. Zhong, M.; Xu, R.; Liao, X.; Zhang, S. Do CSR ratings converge in China? A comparison between RKS and Hexun scores. *Sustainability* **2019**, *11*, 3921. [CrossRef]
- 105. Guerrero-Villegas, J.; Giráldez-Puig, P.; Pérez-Calero Sánchez, L.; Hurtado-González, J.M. Ownership concentration and firm performance: The moderating effect of the monitoring and provision of resources board roles. *Rev. Esp. Financ. Contabilidad* **2018**, 47, 464–484. [CrossRef]
- 106. Kim, H.T.; Kwak, B.; Lee, J.; Suk, I. CEO and outside director equity compensation: Substitutes or complements for management earnings forecasts? *Eur. Account. Rev.* **2019**, *28*, 371–393. [CrossRef]
- 107. Calegari, M.F.; Chotigeat, T.; Harjoto, M.A. Corporate social responsibility and earnings reporting. *J. Curr. Res. Glob. Bus.* **2010**, 13, 1.

Sustainability **2022**, 14, 16208 26 of 27

108. MacKenzie, D.A.; Muniesa, F.; Siu, L. Do Economists Make Markets?: On the Performativity of Economics; Princeton University Press: Princeton, NJ, USA, 2007.

- 109. Huang, W.; Boateng, A. Executive shareholding, compensation, and analyst forecast of Chinese firms. *Appl. Econ.* **2017**, 49, 1459–1472. [CrossRef]
- 110. Mahoney, L.; Roberts, R.W. Corporate social and Environmental Performance and Their Relation to Financial Performance and Institutional Ownership: Empirical Evidence on Canadian Firms. 2022. Available online: https://papers.ssrn.com/sol3/papers.cfm?abstract_id=305781 (accessed on 24 October 2022).
- 111. Akhtaruddin, M.; Haron, H. Board ownership, audit committees' effectiveness and corporate voluntary disclosures. *Asian Rev. Account.* **2010**, *18*, 68–82. [CrossRef]
- 112. Nazari, J.A.; Hrazdil, K.; Mahmoudian, F. Assessing social and environmental performance through narrative complexity in CSR reports. *J. Contemp. Account. Econ.* **2017**, *13*, 166–178. [CrossRef]
- 113. Koh, K.; Li, H.; Tong, Y.H. Corporate social responsibility (CSR) performance and stakeholder engagement: Evidence from the quantity and quality of CSR disclosures. *Corp. Soc. Responsib. Environ. Manag.* **2022**, 1–14. [CrossRef]
- 114. Lai, S.M.; Liu, C.L.; Wang, T. Increased disclosure and investment efficiency. *Asia-Pac. J. Account. Econ.* **2014**, 21, 308–327. [CrossRef]
- 115. Khan, M.A.; Yau, J.T.H.; Marsidi, A.; Ahmed, Z. Pushing a balloon: Does corporate risk disclosure matter for investment efficiency? *J. Financ. Report. Account* ahead of print. **2022**. [CrossRef]
- 116. Cheynel, E. A theory of voluntary disclosure and cost of capital. Rev. Account. Stud. 2013, 18, 987–1020. [CrossRef]
- 117. Borch, O.J.; Huse, M. Informal strategic networks and the board of directors. Entrep. Theory Pract. 1993, 18, 23–36. [CrossRef]
- 118. Patel, P.C.; Li, M.; del Carmen Triana, M.; Park, H.D. Pay dispersion among the top management team and outside directors: Its impact on firm risk and firm performance. *Hum. Resour. Manag.* **2018**, *57*, 177–192. [CrossRef]
- 119. Kim, M.; Boyd, D.E.; Kim, N.; Cheong, H.Y. CMO equity incentive and shareholder value: Moderating role of CMO managerial discretion. *Int. J. Res. Mark.* **2016**, *33*, 725–738. [CrossRef]
- 120. Chang, Y.K.; Oh, W.Y.; Park, J.H.; Jang, M.G. Exploring the relationship between board characteristics and CSR: Empirical evidence from Korea. *J. Bus. Ethics* **2017**, 140, 225–242. [CrossRef]
- 121. Core, J.; Guay, W. The use of equity grants to manage optimal equity incentive levels. *J Account Econ.* **1999**, 28, 151–184. [CrossRef]
- 122. Benmelech, E.; Kandel, E.; Veronesi, P. Stock-based compensation and CEO (dis) incentives. *Q. J. Econ.* **2010**, 125, 1769–1820. [CrossRef]
- 123. Ang, S.H.; Benischke, M.H.; Doh, J.P. The interactions of institutions on foreign market entry mode. *Strateg. Manag. J.* **2015**, *36*, 1536–1553. [CrossRef]
- 124. Lu, J.W. Intra-and inter-organizational imitative behavior: Institutional influences on Japanese firms' entry mode choice. *J. Int. Bus. Stud.* **2002**, 33, 19–37. [CrossRef]
- 125. Panwar, R.; Paul, K.; Nybakk, E.; Hansen, E.; Thompson, D. The legitimacy of CSR actions of publicly traded companies versus family-owned companies. *J. Bus. Ethics* **2014**, *125*, 481–496. [CrossRef]
- 126. Bachmann, P.; Ingenhoff, D. Legitimacy through CSR disclosures? The advantage outweighs the disadvantages. *Public Relat. Rev.* **2016**, 42, 386–394. [CrossRef]
- 127. Aghion, P.; Dechezleprêtre, A.; Hemous, D.; Martin, R.; van Reenen, J. Carbon taxes, path dependency, and directed technical change: Evidence from the auto industry. *J. Polit. Econ.* **2016**, *124*, 1–51. [CrossRef]
- 128. Kang, J. The relationship between corporate diversification and corporate social performance. *Strateg. Manag. J.* **2013**, 34, 94–109. [CrossRef]
- 129. Slangen, A.H. Greenfield or acquisition entry? The roles of policy uncertainty and MNE legitimacy in host countries. *Glob. Strateg. J.* **2013**, *3*, 262–280. [CrossRef]
- 130. Reusen, E.; Stouthuysen, K. Misaligned control: The role of management control system imitation in supply chains. *Account. Organ. Soc.* **2017**, *61*, 22–35. [CrossRef]
- 131. Barreto, I.; Baden-Fuller, C. To conform or to perform? Mimetic behaviour, legitimacy-based groups and performance consequences. *J. Manag. Stud.* **2006**, *43*, 1559–1581. [CrossRef]
- 132. Benischke, M.H.; Martin, G.P.; Gomez-Mejia, L.R.; Ljubownikow, G. The effect of CEO incentives on deviations from institutional norms in foreign market expansion decisions: Behavioral agency and cross-border acquisitions. *Hum. Resour. Manag.* **2020**, *59*, 463–482. [CrossRef]
- 133. Meng, S.; Yan, J.; Cao, X. Heterogeneity in top management teams and outward foreign direct investment: Evidence from Chinese listed companies. Front. *Bus. Res. China* **2019**, *13*, 1–28. [CrossRef]
- 134. Lu, J.; Liu, X.; Filatotchev, I.; Wright, M. The impact of domestic diversification and top management teams on the international diversification of Chinese firms. *Int. Bus. Rev.* **2014**, *23*, 455–467. [CrossRef]
- 135. Hambrick, D.C.; Mason, P.A. Upper echelons: The organization as a reflection of its top managers. *Acad. Manag. Rev.* **1984**, *9*, 193–206. [CrossRef]
- 136. Hallin, C.; Lind, C.H. Revisiting the external impact of MNCs: An empirical study of the mechanisms behind knowledge spillovers from MNCs subsidiaries. *Int. Bus. Rev.* **2012**, *21*, 167–179. [CrossRef]
- 137. Pan, X.; Li, M.; Wang, M.; Chu, J.; Bo, H. The effects of outward foreign direct investment and reverse technology spillover on China's carbon productivity. *Energy Policy* **2020**, *1*45, 111730. [CrossRef]

Sustainability **2022**, 14, 16208 27 of 27

138. Du, L.; Lin, R. OFDI, Reverse Technology Spillovers and Provincial Innovation in China: A Threshold Test Based on Interprovincial Panel Data in China. *China Soft Sci.* **2018**, *1*, 149–162.

- 139. Zhang, G.; Wang, L.; Guo, F.; Yang, G. Does corporate internationalization affect corporate social responsibility? Evidence from China. *Emerg. Mark. Rev.* **2021**, *46*, 100794. [CrossRef]
- 140. Kong, Q.; Tong, X.; Peng, D.; Wong, Z.; Chen, H. How factor market distortions affect OFDI: An explanation based on investment propensity and productivity effects. *Int. Rev. Econ. Financ.* **2021**, *73*, 459–472. [CrossRef]
- 141. Fabrizi, M. Chief marketing officer's equity incentives: Economic determinants and effects on shareholder value. *Eur. J. Market.* **2014**, *48*, 1757–1781. [CrossRef]
- 142. Costca, M.; Torrecchia, P. The concept of value for CSR: A debate drawn from Italian Classical Accounting. *Corp. Soc. Responsib. Environ. Manag.* **2017**, 25, 113–123. [CrossRef]
- 143. Nwoba, A.C.; Boso, N.; Robson, M.J. Corporate sustainability strategies in institutional adversity: Antecedent, outcome, and contingency effects. *Bus. Strateg. Environ.* **2021**, *30*, 787–807. [CrossRef]
- 144. Anh, D.B.H.; Tien, N.H. Gaining competitive advantage from corporate social responsibility policy change. *Int. J. Res. Financ. Manag.* **2019**, *2*, 8–12.
- 145. Xu, Z.; Hou, J. Effects of CEO overseas experience on corporate social responsibility: Evidence from Chinese manufacturing listed companies. *Sustainability* **2021**, *13*, 5335. [CrossRef]
- 146. Sater, D.J.; Dixon-Fowler, H.R. CEO international assignment experience and corporate social performance. *J. Bus. Ethics* **2009**, *89*, 473–489. [CrossRef]
- 147. Derchi, G.B.; Zoni, L.; Dossi, A. Corporate social responsibility performance, incentives, and learning effects. *J. Bus. Ethics* **2021**, 173, 617–641. [CrossRef]
- 148. Kaplan, R.S.; Norton, D.P. *The Balanced Scorecard: Translating Strategy into Action*; Harvard Business School Press: Boston, MA, USA, 1996.
- 149. CSRC Administrative Punishment Decision Letter. No. 36; 2018. Available online: http://www.csrc.gov.cn/pub/zjhpublic/G003 06212/201806/t20180614_339886.htm (accessed on 24 October 2022).